LSE

0

Department of International Development



DIGITAL MENTAL HEALTH AND PSYCHOSOCIAL SUPPORT CHALLENGES AND BEST PRACTICES

> Camille Neidhardt Soledad Palacios Mateo Fabian Schmidt-Gödelitz Loretta Tsang



Psychosocial Centre

Croix-Rouge suisse Schweizerisches Rotes Kreuz Croce Rossa Svizzera



Disclaimer

This report has been prepared for the International Federation of the Red Cross and Red Crescent (IFRC) and the Swiss Red Cross (SRC) by a consultancy team from The London School of Economics and Political Science (LSE). The views expressed are those of the authors and do not reflect those of the IFRC, the SRC, or the LSE.

© March 2022 All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means electronic, mechanical, photocopying, recording, or otherwise, without full attribution.

Table of Contents

Executive Summary	iii
I. Introduction	1
1. Background and Purpose of this document	1
 2. Key Concepts 2.1 Mental Health 2.2 Psychosocial Support 2.3 Mental Health and Psychosocial Support (MHPSS)	2 2 3 3
 3. Methodology 3.1 Desk-Based Research 3.2 Key informant interviews 3.3 Limitations 	3 3 3 4
4. Ethics and Anonymity	4
II. The Approach: Digital MHPSS	7
1. The IRCRCM Approach to MHPSS	7
2. Analytical Framework for Digital MHPSS Interventions: SESSP	8
III. From Conventional to Digital MHPSS	. 11
 Current Landscape of conventional MHPSS within the IRCRCM	11 11 12 14
 2. The Potential of Digital MHPSS 2.1 Safety 2.2 Effectiveness 2.3 Scalability 2.4 Sustainability 2.5 Participation 	15 15 16 17 18 18
3. Current landscape of Digital MHPSS Interventions	20
 IV. Digital MHPSS: Challenges & Solutions. 1. Safety. 1.1 Data Protection	. 23 23 23 25 26 26
2. Effectiveness 2.1 Positive impact 2.2 Acceptance 2.3 Cultural appropriateness	28 28 29 29
 3. Scalability	33 33 35 37
4.1 Temporal Sustainability	37
4.2 Structural Sustainability	38
5. Participation	40
V. Recommendations	. 45

VI. Critical Considerations	51
VII. Conclusion	53
Bibliography	vi
Annexes	xii
Annex 1: Interviewee List	xii
Annex 2: List of Interventions	xiii
Annex 3: Interview Guide	xxiii

List of figures

Figure	1: Overview of interviews conducted	5
Figure	2: The IRCRCM Mental Health and Psychosocial Support Framework	7
Figure	3: SESSP – Analytical Framework for Assessing Digital MHPSS Best Practices	8
Figure	4: Provision of psychosocial support activities	.12
Figure	5: Provision of mental health activities	. 12
Figure	6: MHPSS Budgets	.13
Figure	7: Staff involved in MHPSS activities	.13
Figure	8: Volunteers involved in MHPSS activities	. 13
Figure	9: Classification of current interventions according to pyramid level	.21
Figure	10: Flowchart Illustration of Best Practices in Digital MHPSS	.43
Figure	11: A Hybrid Model for integrated MHPSS programmes	.46

List of tables

Table 1: SESSP – Analytical Framework for Assessing MHPSS Best Practices	9
Table 2: Summary Table of common challenges and viable solutions	42

List of boxes

Box	1: Psychological First Aid	11
Box	2: Fadfada	15
Box	3: Step-by-Step	20
Box	4: Boblberg	24
Box	5: Universal App Program/First Aid App	31
Box	6: SUI-Digital psychological and psychosocial support for refugees	32
Box	7: Remote Assistance Service for the Elderly	39
Box	8: E-referral system during COVID-19	39

Acknowledgements

The consultancy team would like to express its gratitude to all interviewees who took the time to answer our many questions. Without their dedication and passion, this report would not have been possible. We would also like to thank Monia Aebersold from the Swiss Red Cross and Michelle Engels from the IFRC Reference Centre for Psychosocial Support for enabling this project and for their patient guidance. Finally, we would like to thank Prof. Jean-Paul Faguet for his mentorship throughout the process.

List of Acronyms / Glossary

GDPC Global Disaster Preparedness Center Reference centre to support innovation and learning in disaster preparedness, established by the American Red Cross and the International Federation of the Red Cross and Red Crescent. IASC Inter-Agency Standing Committee The highest-level coordination forum of the United Nations system for humanitarian action. It regroups 18 international organisations inside and outside the UN to develop policies, set priorities of action and mobilise resources for humanitarian action. ICRC International Committee of the Red Cross Independent humanitarian organisation mandated to protect people in armed conflict worldwide. IFRC International Federation of Red Cross and Red Crescent Societies Global Secretariat of the National Red Cross Red Crescent Societies IFRC PS International Federation of the Red Cross and Red Crescent Reference Centre for Centre **Psychosocial Support** Reference centre of the International Federation of the Red Cross and Red Crescent. mandated to support National Societies in promoting and enabling the psychosocial well-being of affected populations, staff, and volunteers. IRCRCM International Red Cross Red Crescent Movement The overarching international humanitarian movement which regroups the individual National Red Cross Red Crescent Societies, the International Federation of Red Cross and Red Crescent Societies and the International Committee of the Red Cross. It is made up of nearly 14 million volunteers and over 160,000 local branches in 192 countries. MHPSS Mental Health and Psychosocial Support

According to the Inter-Agency Standing Committee, it is 'a composite term for any type of local or outside support that aims to protect or promote psychosocial well-being and/or prevent or treat mental disorder'.

NS	National Red Cross Red Crescent Society Autonomous national organisation committed to the Fundamental Principles of the International Red Cross Red Crescent Movement. Each of the 192 National Societies consists of a network of local volunteers and staff who provide a wide variety of services.		
PFA	Psychological First Aid A method designed for emergency intervention, providing short-term relief to people experiencing an acute mental health crisis.		
RCT	Randomised Controlled Trial An experimental evaluation method based on the random attribution of a treatment and controlling of extraneous variables between the treatment and control groups to evaluate the efficacy of a treatment.		
SOP	Standard Operating Procedure		
UNDP	United Nations Development Program United Nations agency dedicated to advancing development and eradicating poverty worldwide.		
UNHCR	United Nations High Commissioner for Refugees United Nations Refugee Agency dedicated to protecting rights and lives of refugees, forcibly displaced and stateless people.		
UNICEF	United Nations International Children's Emergency Fund United Nations agency dedicated to providing humanitarian aid and development assistance to children worldwide.		
WHO	World Health Organization United Nations agency that connects key stakeholders in promoting health and fighting diseases.		

Executive Summary

Addressing mental health and psychosocial wellbeing has become imperative worldwide, however, the treatment gap is only increasing. This is especially the case for low- and middle-income countries, societies affected by emergencies and conflicts, and vulnerable groups. This report aims at recommending ways in which digital technologies could help address this gap effectively. This report is structured around seven Chapters:

Chapter I introduces the purpose and research scope of this report before presenting the key concepts with which this report engages, and the methodology employed. Commissioned by the IFRC Reference Centre for Psychosocial Support and the Swiss Red Cross in partnership with the London School of Economics and Political Science (LSE), this study analyses how digital technologies can be leveraged for improving the provision of mental health and psychosocial support (MHPSS) services and the challenges faced by practitioners throughout this process. Importantly, the report aims at identifying solutions from the field and proposing practical recommendations to address these challenges. To achieve so, 32 interviews with key informants were conducted, complemented with in-depth desk-based research.

Chapter II presents the theoretical approach to digital MHPSS adopted in this report. After introducing the International Red Cross and Red Crescent Movement (IRCRCM) approach to MHPSS, the report proposes an analytical framework to assess digital MHPSS practices. The framework is composed of the five main elements – safety, effectiveness, scalability, sustainability, and participation (SESSP) – and is devised to guide the design, implementation, and evaluation of digital MHPSS interventions.

Chapter III considers how digital technologies can advance the provision of MHPSS services. The chapter analyses the current landscape of conventional MHPSS activities within the Movement. It is argued that digital technologies have the potential of overcoming some of the specific needs encountered by National Societies in the provision of conventional MHPSS services to date. This is because i) digital interventions can overcome the barrier of stigma thanks to anonymity; ii) digital interventions can be equally effective as in-person interventions, in certain circumstances; iii) digital interventions allow scaling up services access to previously inaccessible populations, and at minimal marginal costs; iv) digital technologies can improve the continuum of care if integrated into the wider care system and can improve local capacity-building; and v) digital interventions can make MHPSS provision more participatory. In short, digital MHPSS has the potential of bridging the growing treatment gap in MHPSS. Finally, this chapter also gives an overview of how digital MHPSS is used in the Movement so far.

Chapter IV presents the evidence collected through the interviews. The section progresses following the SESSP Analytical Framework, systematically matching common challenges faced by practitioners at each level to practical solutions emanating from field experience with digital MHPSS. Main themes include data security, quality of care, cultural appropriateness, resource limitations, accessibility of services, and inclusiveness.

Chapter V synthesises overarching solutions from the field and from practical recommendations that arose from the previous analysis of challenges and solutions. The identified recommendations are intended to advance the provisions of MHPSS services by pointing out how and when can digital technologies enhance this provision. The recommendations are:

- 1. To the Movement: Leverage digital tools to collect data on digital interventions for monitoring and evaluation. Leverage the data thus gathered to enhance planning.
- 2. To the Movement: Develop hybrid models of both digital and conventional MHPSS within single interventions and comprehensive programs to maximise efficiency and impact.
- a) To the Movement: Adopt and adapt existing evidence-based interventions
 b) To the IFRC: Facilitate adoption by creating, updating, and disseminating a whitelist of evidence-based interventions; and a blacklist of unsafe interventions
 - c) To the IFRC: Create an MHPSS Innovation centre to develop, maintain and update new interventions to be distributed at minimal marginal cost
- 4. To the Movement: Critically engage in partnerships to overcome resource gaps and optimise outcomes. Partnerships with private companies must comply with special safety standards to ensure the integrity of humanitarian principles.
- 5. To the Movement: Build synergies with local structures by organising efficient task transfers and referrals, and by building local capacities.
- 6. To the IFRC: Develop digital training modules, Standard Operating Procedures, and supervision for practitioners to improve quality of care and to safeguard practitioners' wellbeing.

Chapter VI highlights three critical considerations that should be accounted for when providing digital MHSPP services.

- 1. First, there is a bias in the literature on the effectiveness of digital MHPSS interventions, as most of the research has been conducted in Western and high-resource contexts. Hence the evidence base of digital MHPSS interventions should be critically considered.
- 2. Second, the inequitable access to digital MHPSS should be considered. The capacity to digitalize MHPSS services differs between the Global North and the Global South. In addition, other factors such as gender, also affect individuals' access to digital MHPSS services.
- Finally, digital MHPSS interventions are just one aspect in the advancement of individual and social wellbeing. Without an enabling environment for holistic wellbeing, the impact of digital MHPSS services is limited.

Chapter VII concludes by identifying areas of further research and calling upon the wider MHPSS community to create a durable platform for knowledge sharing and innovation in digital MHPSS.



I. INTRODUCTION



I. Introduction

1. Background and Purpose of this document

Nearly one-third of the global population suffers from a common mental disorder at one point in their lifetime (primarily mood, anxiety, and substance use disorders).¹ The ongoing COVID-19 pandemic has further deteriorated global mental wellbeing,² and disrupted critical mental health services worldwide.³

Treating and preventing mental health problems is an urgent concern for global health. Apart from stress and mental illness being a risk factor for many diseases, globally more than 700,000 lives are lost to suicide every year.⁴ Mental health problems also hamper productivity, with the approximate global cost of 6 trillion USD by 2030.⁵ Most importantly, however, mental wellbeing is a human right and a cornerstone of human flourishing.⁶

However, the world faces a pervasive provision gap in mental health and psychosocial support (MHPSS), underpinned by a chronic shortage of mental health staff. The global median number of mental health workers stands at 13 per 100,000 inhabitants and only 1.4 and 3.8 in lower- and lower-middle-income countries, respectively.⁷ The IRCRCM as one of the most important providers of humanitarian MHPSS faces additional challenges as it is routinely engaged in circumstances where provision and support work are marred by access difficulties, like natural disasters or armed conflict. These challenges create an urgent need for innovative solutions that provide adequate care to affected populations and alleviate the burden off service providers. Digital technologies have emerged as a promising approach to overcoming this provision gap by bridging physical distances and serving larger populations at lower cost.

To advance the provision of MHPSS, the IRCRCM adopted a resolution in 2017 aiming at strengthening the Movement's collective response to mental health and psychosocial needs. In 2019, the Movement's 'Policy on Addressing Mental Health and Psychosocial Needs' was published, reiterating the Movement's approach to MHPSS, and establishing eight policy statements,⁸ together with a Roadmap for 2020-2023 for achieving set targets. Given the potential of digital interventions in improving MHPSS

¹ Steel et al., "The Global Prevalence Of Common Mental Disorders: A Systematic Review And Meta-Analysis 1980– 2013."

² Santomauro et al., "Global Prevalence And Burden Of Depressive And Anxiety Disorders In 204 Countries And Territories In 2020 Due To The COVID-19 Pandemic"

³ WHO, "The Impact Of COVID-19 On Mental, Neurological And Substance Use Services."

⁴ WHO, "Suicide."

⁵ The Lancet Global Health, "Mental Health Matters."

⁶ Patel and Saxena, "Transforming Lives, Enhancing Communities — Innovations In Global Mental Health."; Patel et al., "The Lancet Commission On Global Mental Health And Sustainable Development."

⁷ WHO, Mental Health Atlas 2020.

⁸ The Working Group of the International Red Cross and Red Crescent Movement Project on Addressing Mental Health and Psychosocial Consequences of Armed Conflicts, Natural Disasters and Other Emergencies, *International Red Cross And Red Crescent Movement Policy On Addressing Mental Health And Psychosocial Needs*.

delivery, the Roadmap encourages the design, evaluation, and documentation of digital MHPSS delivery to facilitate the achievement of the Movement's MHPSS goals by the end of 2023.⁹ The present report aims to contribute to this task by answering **four research questions**:

- 1. What digital MHPSS interventions exist inside and outside the Movement?
- 2. What are the common challenges faced by actors inside and outside of the Movement in their implementation of digital MHPSS?
- 3. What are viable solutions and best practices in overcoming these challenges?
- 4. What should relevant stakeholders do to effectively leverage digital innovations for MHPSS within the Movement?

This report aims to present an analysis of what practices exist today and practical solutions for addressing the challenges seen in the field. The report proceeds as follows: First, conceptual and methodological issues will be clarified. Second, the report reviews the IRCRCM's MHPSS approach and devises an analytical framework on digital MHPSS. Based on a needs assessment of current MHPSS within the Movement, the report highlights the potential value added by digitalising MHPSS interventions. Third, the report presents the findings from desk-based research and key informant interviews, presenting an overview of existing digital MHPSS, summarising common challenges and highlighting viable solutions. Finally, a set of six recommendations will be derived, before engaging in a critical reflection of the analysis.

2. Key Concepts

2.1 Mental Health

According to the World Health Organisation (WHO), health encompasses wellbeing in physical, mental and social aspects, where mental health is a 'state of well-being in which the individual realises his or her own abilities, can cope with the normal stresses of life, can work productively and fruitfully, and is able to make a contribution to his or her community'.¹⁰

2.2 Psychosocial Support

In explaining the causes of mental health conditions, social factors are just as important as individual factors, if not more. As suggested in the Bronfenbrenner Ecological Systems Theory, ¹¹ individual development is affected by the surrounding context, from microsystems such as family and school to macrosystems such as societal norms. If well-aligned, these social systems can be a crucial source of psychological resilience.¹²

⁹ International Red Cross And Red Crescent Movement Commitments On Addressing Mental Health And Psychosocial Needs: A Roadmap For Implementation 2020 – 2023.

¹⁰ "Mental Health: Strengthening Our Response."

¹¹ Bronfenbrenner, *The ecology of human development: Experiments by nature and design*.

¹² International Committee of The Red Cross, *Guidelines On Mental Health And Psychosocial Support*.

2.3 Mental Health and Psychosocial Support (MHPSS)

According to the Inter-Agency Standing Committee, MHPSS is a 'composite term [...]to describe any type of local or outside support that aims to protect or promote psychosocial well-being and/or prevent or treat mental disorder'.¹³ MHPSS encompasses multi-layered responses that can address mental health needs in the short- and long term. MHPSS interventions are situated along a continuum ranging from general to focused intervention, from community-focused to individual-focused.¹⁴

2.4 Digital MHPSS

Digital mental health (or 'e-mental health') describes 'the use of information and communication technology (ICT) – in particular the many technologies related to the Internet – to support and improve mental health conditions and mental health care.' ¹⁵ Accordingly, digital MHPSS refers to MHPSS interventions that are supported by ICTs.

3. Methodology

To address the research questions, we employ a two-part mixed methods approach. At its core, we conducted 32 key informant interviews from which we derived information on 37 digital MHPSS interventions. This analysis is complemented by desk-based research.

3.1 Desk-Based Research

To establish a theoretical foundation for our research, we conducted desk-based research on the effectiveness of digital and conventional MHPSS and reviewed existing guidelines on MHPSS and digital technologies in humanitarian settings, drawing on scientific publications and grey literature of various humanitarian organisations. This research highlights key issues and debates around (digital) MHPSS, pinpoints challenges and current developments, and informs the Analytical Framework against which we evaluate our findings. Moreover, web-based research allows us to identify existing digital MHPSS interventions inside and outside the movement.

3.2 Key informant interviews

The core part of our data collection rests on the realisation of 32 semi-structured key informant interviews with mental health experts and field practitioners from inside (20) and outside the Movement (12) (cf. Annex 1 for the full list of interviewees) who reported their experiences in implementing 37 digital MHPSS interventions (cf. Annex 2). Semi-structured interviews are the best-suited data collection method for our study because they allow a deeper understanding of digital MHPSS in a purposive manner while giving interviewees ample freedom to express what they deem important.¹⁶

¹³ Inter-Agency Standing Committee, IASC Guidelines On Mental Health And Psychosocial Support In Emergency Settings.

¹⁴ Lee et al., "Identifying Research Priorities For Psychosocial Support Programs In Humanitarian Settings."

¹⁵ Riper et al., "Theme Issue On E-Mental Health: A Growing Field In Internet Research."

¹⁶ Hamilton and Corbett-Whittier, Using Case Study In Education Research.

Key informants were initially identified using purposive sampling, in close consultation with the clients. This method is appropriate because it allows us to identify informants who were most likely to provide relevant insights.¹⁷ The selection was made based on the following criteria: professional experience with digital MHPSS; balance between researchers and practitioners; geographical balance; balance between internal and external informants. We complemented this initial sample via a snowball sampling method, eliciting the knowledge of our interviewees to identify more knowledgeable experts on the topic. Moreover, one participant observation in an internal workshop by the Movement's Global Disaster Preparedness Centre was conducted.

3.3 Limitations

First, privileging in-depth informant interviews instead of surveys reduces the breadth of the data. Second, our sampling method did not allow for a representative sample of interviewees, leading to an under-representation of lower-income regions that reflects structural disadvantages in digitalising MHPSS. Third, though the team is international in make-up, we are deeply aware of our biases in analysing social and mental health issues from across different contexts.

4. Ethics and Anonymity

Our empirical methodology was approved by the LSE Ethics Committee. We elicited the informed consent of all interviewees via signed consent forms and included direct quotes only upon explicit authorisation.



¹⁷ Tongco, "Purposive Sampling As A Tool For Informant Selection."



Figure 1: Overview of interviews conducted¹⁸

¹⁸ Authors' own production.



II. The approach: Digital mhpss

II. The Approach: Digital MHPSS

1. The IRCRCM Approach to MHPSS

The IRCRCM's MHPSS model consists of a four-layered intervention pyramid (Figure 2). As interventions move up the pyramid from basic psychosocial support to specialised mental health care, their focus moves from promotion of mental wellbeing to prevention of mental health conditions, to treating distress and mental health conditions. The higher up the pyramid, the more focused and specialised care becomes, and the higher levels of competencies and training are required. This continuum of care should be situated in a protective environment where rights and dignity of service users are respected. This report adopts this approach throughout its considerations on digital MHPSS.¹⁹



Figure 2: The IRCRCM Mental Health and Psychosocial Support Framework²⁰

¹⁹ Psychosocial Centre International Federation of Red Cross and Red Crescent Societies, "The MHPSS Framework."
²⁰ Adapted from Psychosocial Centre International Federation of Red Cross and Red Crescent Societies, "The MHPSS Framework."

2. Analytical Framework for Digital MHPSS Interventions: SESSP

To assess practices of digital MHPSS and guide corresponding recommendations, this report establishes an original analytical framework encompassing five key dimensions: **Safety, Effectiveness, Scalability, Sustainability,** and **Participation**. Each dimension consists of several facets which highlight key determinants of the success of digital interventions (Figure 3 and Table 1).

This set of guiding principles is developed based on the 'Guidelines on Mental Health and Psychosocial Support' by the ICRC,²¹ as well as the 'Common Monitoring and Evaluation Framework for Field test version Mental Health and Psychosocial Support in Emergency Settings' developed by Inter-Agency Standing Committee (IASC),²² the 'Monitoring and Evaluation Framework for Psychosocial Support Interventions' in the Movement,²³ and the 'Mental Health And Psychosocial Technical Note',²⁴ in addition to practical concerns and needs of the IRCRCM. This analytical framework is not meant to be exhaustive, but to serve as a guide to highlight core principles in evaluating and recommending digital MHPSS.



Figure 3: SESSP – Analytical Framework for Assessing Digital MHPSS Best Practices²⁵

²¹ The 5 principles are: (1) Upholding humanity, impartiality, and non-discrimination, (2) Ensuring community participation and cultural awareness, (3) Complying with medical and health care ethics, (4) Delivering quality of care in line with internationally accepted standards, (5) Ensuring a continuum of care, International Committee of The Red Cross, *Guidelines On Mental Health And Psychosocial Support*.

 ²² IASC Reference Group on Mental Health and Psychosocial Support in Emergency Settings, *Mental Health And Psychosocial Support In Emergency Settings - Monitoring And Evaluation With Means Of Verification: Version 2.0.* ²³ Psychosocial Centre International Federation of Red Cross and Red Crescent Societies, *IFRC Monitoring And Evaluation Framework For Psychosocial Support Interventions*.

²⁴ UNICEF, Mental Health And Psychosocial Technical Note.

²⁵ Authors' own production.

Dimension	Facet	Indicative questions
	Data Protection	 Is data securely protected with anonymity? Is data used and accessed by designated parties for appropriate use?
Safety	Do No Harm	 Does the intervention have adverse impacts on users? How are possible adverse impacts/ unsafe use of users monitored? Are the staff properly trained by professionals, following a standard operating protocol that outlines ethical principles in implementing and using the intervention?
,	Crisis Intervention	 Are there crises intervention protocols and systems that promptly address emergencies (such as acute distress, sui- cidality) that users face in digital MHPSS tools?
	Practitioners' methods and wellbeing	 Are the practitioners' methods impeded by the digital in some ways? Is the practitioners' wellbeing being taken care of considering the risks they are exposed to as they adopt digital MHPSS tools addressed?
	Positive effect on MHPSS	 Are primary/ secondary outcomes (symptom reduction, prevention, improve quality of life) achieved (as measured and evaluated by reliable and validated tests)? Do outcomes last in the long run?
Effectiveness	Acceptance of the inter- vention by the target pop- ulation	Is the intervention well-received by the target population?Does it respond to specific challenges/needs of the target population?
	Cultural Appropriateness	How well does the intervention adapt to local contexts?Is it relevant (culturally, socially) for the target population?
Saalahility	Accessibility	 How far can the intervention reach a population with dif- ferent characteristics (such as literacy level, de- mographics)?
	Resource Limitations	 Does it require a high level of technical capability/ financial and human capital? Does it require staff with a high level of specialisation?
Sustainability	Temporal sustainability	 Could long-term funding be secured? How sustainable is the partnership for the intervention? How sustainable are the human resources in implementing the intervention?
	Structural sustainability	 Is the intervention embedded into the wider healthcare system and work in synergy with existing infrastructure (e.g., governmental and professional bodies)
	Diversity	 Are differences and intersectionalities within user groups respected reflected in and considered in the design and im- plementation processes?
Participation	Inclusion	 Are user groups meaningfully involved in the design, implementation, and evaluation? Are users' feedback captured by evaluation mechanisms and addressed?

Table 1: SESSP – Analytical Framework for Assessing MHPSS Best Practices



III. FROM CONVENTIONAL TO DIGITAL MHPSS

1

III. From Conventional to Digital MHPSS

This chapter gives an overview of current conventional MHPSS interventions within the Movement to identify needs to be addressed. It then explores the potential of digital MHPSS in overcoming these needs. Finally, it presents an overview of how digital MHPSS is used in the Movement so far.

1. Current Landscape of conventional MHPSS within the IRCRCM

1.1 Scope of Activities

In October 2021, the Movement published the report 'Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities', based on a survey of 163 National Red Cross Red Crescent Societies (hereinafter NS).²⁶ The report compares the Movement's (conventional) MHPSS activities from 2021 with those conducted in 2019. From the analysis of the survey data, we can derive current challenges and needs related to MHPSS in the Movement.

According to the survey, approximately 94% of all NS provide some MHPSS services, while 83% have included an MHPSS focus within their strategy.

Psychosocial interventions are generally more common than mental health interventions. 59% of NS conduct stress management activities, 54% conduct self-care, 51% psychoeducation and 46% conduct anti-stigma interventions, among a large variety of activities. **Psychological First Aid** skills are strongly diffused among staff and volunteers across all regions. Mental health interventions are less common and focus on lower-bound psychological support (63%), group therapy and support groups (35%), and training in basic psychological support (48%). However, 70% of NS offer referrals to more specialised mental health services (e.g., psychiatrists and psychologists).



Psychological First Aid

Psychological First Aid (PFA) is designed as an immediate response to crises, stabilising and assessing needs of people affected.²⁷ It has been widely endorsed and integrated into MHPSS response structures for disasters.²⁸ PFA aims at facilitating long-term recovery by helping service users feel safe, connected, accessing support (social, physical, emotional) and feeling capable of helping themselves.²⁹ The key action principles of PFA are Look, Listen and Link.³⁰ Studies have cautioned that while PFA allowed greater accessibility of basic MHPSS, it was also used to substitute higher-level or other MHPSS in some instances.³¹

For more information: 'A Guide to Psychological First Aid for Red Cross and Red Crescent Societies' by Psychosocial Centre, IFRC

²⁶ IRCRCM, "Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities".

²⁷ American Psychological Association. "Understanding Psychological First Aid".

²⁸ Shultz and Forbes, "Psychological First Aid."

²⁹ WHO. Psychological First Aid: Guide For Field Workers.

³⁰ Psychosocial Centre IFRC, A Guide to Psychological First Aid for Red Cross and Red Crescent Societies.

³¹ Snider, *Psychological First Aid: Five Year Retrospective (2011-2016).*



Figure 4: Provision of psychosocial support activities³²

Figure 5: Provision of mental health activities³³

1.2 Needs and Challenges

The following figures display the most prominent challenges among the NS when implementing MHPSS programmes:

1.2.1. Data Collection and Evaluation

Evaluation of activities is limited in breadth and depth, due to a lack of funds, tools, and staff expertise. Only 10% are involved in MHPSS research and less than 50% of NS report having an information system in place that ensures confidentiality and protection of personal data. The lack of systematic data collection impedes the development of the field, especially since the evidence is thinner for lower-level psychosocial interventions compared to specialised mental health care.

 ³² IRCRCM, "Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities", 11.
 ³³ Ibid, 14.

1.2.2 Financial needs



Financial limitations present the biggest challenge to implementing MHPSS, cited by 75% of NS.

Across regions, 25% of NS report having no budget for MHPSS. $^{\rm 34}$

Of those who are aware of the budget (22% don't know), 19% have less than 50k CHF; 7% have 50-100k, 6% have 100-150k and 7% have 150-200k.

1.2.3 Lack of human resources

Lack of human resources and expertise is a crucial challenge to 42% of NS. Approximately one-third of NS have less than five MHPSS staff members, two thirds have between less than twenty. But there is significant heterogeneity: 12% of NS have more than 100 staff members. Staff limitations are partially balanced by the fact that 37% of NS can mobilise more than 100 volunteers involved in MHPSS. However, the great majority of MHPSS staff and volunteers do not have specialised qualifications.



 ³⁴ According to the publication, this "may be due to the fact that many activities are delivered as an integrated approach and therefore the budget is not captured specifically under MHPSS but is included in other sectors."
 ³⁵ IRCRCM, "Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities", 24
 ³⁶ IRCRCM, "Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities", 20.

1.2.4 Lack of partnerships

Strategies to overcome funding and skills gaps through partnerships are limited by a perceived lack of coordination with other NGOs and governmental organisations involved in MHPSS by NS (38%). There are few collaborations overall; most occur within the Movement, or with national governments. There are only very few partnerships with universities, private businesses, or UN agencies in the field of MHPSS. This stands in contrast to the perceived need for more partnerships in MHPSS, felt by 97% of survey responders. The most prominent need to be addressed through partnerships is that of technical support (78%). Partnerships are hampered by the above-cited limitations in finance and staff, but also logistical difficulties and organisational incompatibilities.

1.2.5 Insufficient access

Organisations face a dual problem of stigmatisation (35%) and limited access to affected individuals (22%), strongly inhibiting their work.

1.3 Conclusion

Overall, it must be noted that the Movement is substantially increasing its MHPSS efforts. However, lack of funding and staff inhibit progress for the implementation of the 2020-2023 MHPSS Roadmap. Prominent shortcomings also include lack of specialised staff, programme evaluation, partnerships, and access to the affected population.



2. The Potential of Digital MHPSS

Considering the structural needs of the Movement identified above, digital delivery formats promise to enhance MHPSS services. In 2019, there were 5.2 billion unique mobile subscribers, 4.1 billion internet users, 3.8 billion active social media users and 204 billion apps downloaded.³⁷ As the digital infrastructure continues to expand, the Movement could build on existing technologies as well as recent innovations – catalysed by the COVID-19 pandemic – to diversify and increase its MHPSS efforts. This section surveys the existing literature to understand where the main benefits of digital MHPSS lie for each of the SESSP principles.

2.1 Safety

Digital interventions can help **overcome stigma** thus reducing barriers to seeking psychological and psychosocial help. Digital tools can be used with full anonymity, allowing individuals to share their experiences without fear of stigmatisation. This is especially important for those who face **'double stigma'**³⁸ such as ethnic minorities, women, the LGBTQ+ population or refugees.



³⁷ Budd et al., "Digital Technologies In The Public-Health Response To COVID-19."

³⁸ Gary, "Stigma: Barrier To Mental Health Care Among Ethnic Minorities."

³⁹ Interviewee 4.

2.2 Effectiveness

Although there is still a need for further research, scientific studies demonstrate strong and significant effects of digital interventions in specific areas. These include remote psychotherapy against depression,⁴⁰ substance use disorder,⁴¹ and anxiety,⁴² among others.⁴³

Furthermore, there is increasing evidence on the effectiveness of automated (self-guided) interventions that address common mental health problems. Self-guided digital interventions, such as **mobile applications based on cognitive-behavioural therapy (CBT)** have been clinically proven to reduce severity in cases of depression, anxiety, insomnia, general stress, and worry.⁴⁴

The effectiveness of remote MHPSS interventions depends greatly on the acceptance of the programme by the service users. Reassuringly, scientific studies find broad **acceptability of evidence-based digital** interventions for mild to moderate mental health problems.⁴⁵

Guided v. Self-guided interventions

Guided digital interventions: can range from audio/video counselling - in which in-person counselling is replicated through digital technologies - to apps and websites that include some kind of interaction with a specialist (e.g., weekly calls/chats). The distinguishing factor behind this intervention is that a counsellor/specialist can be approached by the user when needed, either remotely or in-person.

Self-guided digital interventions: many apps and websites offer services which are largely automated, with the user either self-guiding the intervention or sent automated remainders and notifications related to their therapeutical needs. In this case, the user does not have access to a specialist. Some interventions also combine self-guidance elements with support from practitioners.

⁴⁰ Lungu et al., "Blended Care-Cognitive Behavioral Therapy For Depression And Anxiety In Real-World Settings: Pragmatic Retrospective Study".

⁴¹ Lin et al., "Telemedicine-Delivered Treatment Interventions For Substance Use Disorders: A Systematic Review".

⁴² Berryhill et al., "Videoconferencing Psychological Therapy And Anxiety: A Systematic Review".

⁴³ Hubley et al., "Review Of Key Telepsychiatry Outcomes", 269.

⁴⁴ Attridge, "Internet-Based Cognitive-Behavioral Therapy For Employees With Anxiety, Depression, Social Phobia, Or Insomnia: Clinical And Work Outcomes"; Venkatesan et al., "Digital Cognitive Behavior Therapy Intervention For Depression And Anxiety: Retrospective Study".

⁴⁵ Schneider et al., "Acceptability Of Online Self-Help To People With Depression: Users' Views Of Moodgym Versus Informational Websites".

2.3 Scalability

In contexts in which resources are scarce, access is limited, and gaps in service provision are high, digital MHPSS interventions are efficient alternatives to face-to-face interventions. One of the main advantages of self-guided and remotely guided digital MHPSS interventions is their ability to reach larger parts of the population with fewer human resources. Thus, digital services promise to **bridge the growing treatment gap in MHPSS**.⁴⁶ This is especially important for low and middle-income countries, where the resources are less abundant.⁴⁷

Digital MHPSS has the potential to **improve access and reduce costs** for geographically remote or conflict-affected areas, ⁴⁸ as well as for populations facing other barriers to access, such as transportation costs, lack of free time, lack of childcare, or stigma.⁴⁹ Hence, digital MHPSS has the potential **to lower economic and social costs** associated with accessing MHPSS services. Moreover, digital MHPSS allows for **flexible engagement on-demand and just-in-time**, delivered at individual pace.⁵⁰

In addition, digital MHPSS can reduce the provider's operating costs. Even though designing, implementing, and evaluating digital MHPSS interventions might require a large initial investment, self-guided tools, in particular, allow service provision at **near-zero marginal cost per additional user.**⁵¹

Furthermore, digital tools allow for the development of remotely assisted forms of MHPSS that can be **delivered by non-specialists.** For instance, Self-Help-Plus, developed by the WHO to increase stress management capabilities can be conducted by 'briefly trained lay facilitators'.⁵² Similarly, digital MHPSS allows for the scaled-up diffusion of skills such as PFA, which can be delivered by virtually everyone. By increasing the capacities of communities and individuals to self-care and peer support in bottom-layer MHPSS, community resilience is strongly increased, while simultaneously **enhancing task transfer** as more specialist capacities are freed up for top-tier MHPSS interventions⁵³.

In short, digital technologies can maximise the accessibility of MHPSS services at often similar levels of effectiveness, thus bridging the pervasive treatment gap in MHPSS.

⁴⁶ Lal, "E-Mental Health: Promising Advancements In Policy, Research, And Practice", 57.

⁴⁷ Carter et al., "The Emergence Of Digital Mental Health In Low-Income And Middle-Income Countries: A Review Of Recent Advances And Implications For The Treatment And Prevention Of Mental Disorders".

⁴⁸ Lal and Adair, "E-Mental Health: A Rapid Review Of The Literature", 26.

⁴⁹ Ibid.

⁵⁰ Ibid.

⁵¹ Aebersold, "E-Mental-Health für traumatisierte Geflüchtete. Bericht zum Schritt I: Literatur- und Marktanalyse von erprobten und bewährten Materialien und Ansätzen".

⁵² Brown et al., "Self Help Plus: Study Protocol For A Cluster-Randomised Controlled Trial Of Guided Self-Help With South Sudanese Refugee Women In Uganda", See Annex 2 for further details on Self Help Plus.

⁵³ Aebersold, "E-Mental-Health für traumatisierte Geflüchtete. Bericht zum Schritt I: Literatur- und Marktanalyse von erprobten und bewährten Materialien und Ansätzen", 19.

2.4 Sustainability

Once the technology is put in place, digital interventions often are more financially sustainable, since they have lower variable costs, e.g., for wages.⁵⁴ Moreover, digital tools can improve treatment at early stages, **preventing** the development and aggravation of symptoms that require more expensive treatment.⁵⁵

When digital services are integrated into the wider healthcare system, they can improve the **continuity of care** by offering flexibly available services on-demand, either as stand-alone or awaiting face-to-face treatment and in-between sessions. This is especially important considering the high rates of disengagement associated with mental health services.⁵⁶

Finally, digital tools can enhance capacity building to achieved sustained **community resilience**. First, it is possible to scale-up online training of local lay-implementers of psychosocial interventions. Mobile apps or video call technology bridge the geographical distance between participants and trainers, thus reducing the cost and increasing the accessibility of the workshops⁵⁷. Second, the Internet and social media allow increasing awareness and mainstreaming mental wellbeing support to counteract stigmatisation.⁵⁸ This might strongly improve mental health outcomes for entire communities in the long run.⁵⁹

2.5 Participation

Digital technologies offer the opportunity to create broad-scale interventions that allow **representation and community participation.** To many, digital interventions are a means to democratise humanitarian action.⁶⁰ Social media for example allows people to easily connect to each other in a disaster to provide psychosocial support. Digital technologies can be leveraged to **integrate the target group** in project design (e.g., online focus groups), implementation (capacity-building), monitoring, and evaluation (online user feedback).

⁵⁴ Christensen, Griffiths and Evans, *E-Mental Health In Australia: Implications Of The Internet And Related Technologies For Policy*, 4.

⁵⁵ Ibid.

⁵⁶ O'Brien, Fahmy and Singh, "Disengagement From Mental Health Services."

⁵⁷ Interviewee 21.

⁵⁸ Christensen, Griffiths and Evans, *E-Mental Health In Australia: Implications Of The Internet And Related Technologies For Policy*, 4.

⁵⁹ Ibid.

⁶⁰ Meier, Digital Humanitarians.

Strengths and potential benefits of digital MHPSS interventions

Safety

- Stigmatisation can be avoided thanks to anonymity

Effectiveness

 Scientific evidence proves effectiveness of both guided and unguided digital interventions. Further research is needed.

Scalability

- Improve access for geographically remote populations and populations with other barrier to access
- Flexibility for the user
- Reduce the provider's operating costs

Sustainability

- Reduce burden on mental health specialists as they can be delivered by trained non-specialists
- Improve prevention
- Improve continuity of care (through its integration into the wider health care system)
- Enhance capacity building

Participation

- Improve service user engagement in design, implementation, and evaluation.
- Democratisation of MHPSS services



3. Current landscape of Digital MHPSS Interventions

Based on 32 interviews with researchers and practitioners, we gathered detailed information on 24 digital MHPSS interventions from within the Movement, in addition to 13 interventions from outside the Movement, which might be adaptable by the Movement.

In the upper bounds of the pyramid, the most common use of technology is video calls to conduct remote personal counselling and, in some cases, psychiatric treatment. But there is also a growing body of mobile apps for psychological counselling using CBT-based approaches, which have a strong evidence base (for example, Step-by-Step, SAHA, Sui, Stay fine – see Annex 2).

In general, digital technologies are currently more often employed for psychosocial support, in more diverse forms and with higher levels of automation (self-guidance). For a detailed overview of the reviewed interventions, refer to Annex 2.

Step-by-Step⁶³

WHO in collaboration with multiple partners, including the National Mental Health Programme (NMHP) at the Ministry of Public Health (MoPH) in Lebanon, Freie Universität Berlin, and University of Zurich developed an internet-based guided self-help intervention for adults with depression called Step-by-Step.

It is currently being used in research trials and at scale in Lebanon as both an app and a web-based intervention which guides users through several training modules in psychoeducation, behavioural activation, stress management and increasing social support, using an illustrated narrative approach. Step-by-Step was designed as a scalable treatment for depression in populations affected by adversity. It has been tested in randomised controlled trials with populations in Lebanon, including Syrian refugees, as well as Syrian refugees in Germany, Sweden, and Egypt under the European Union funded STRENGTHS project.

It can be delivered with or without guidance. In the guided version, users access support from trained helpers (e.g., university graduates) who answer questions and review activities, at an individual time investment of 15 - 20 minutes per week. In the contact-on-demand approach there is an in-built messenger to reach e-volunteers.

The intervention was found to be effective in reducing depressive symptoms and positively received in two fully powered RCTs in Lebanon, with more trials underway as part of the STREGNTHS trial. WHO is developing an open source platform for running the intervention and plans to release this and all content under Creative Commons for Step-by-Step for adaptation and use in 2022. The app has been translated and adapted for other user groups such as Albanian migrants in Switzerland and Germany, university students in China and overseas Filipino workers.

For more information, contact <u>psych_interventions@who.int</u> or visit: <u>http://strengths-project.eu/en/strengths-home/</u>

⁶¹ Harper Shehadeh et al., "Step-By-Step, An E-Mental Health Intervention For Depression: A Mixed Methods Pilot Study From Lebanon."

⁶² van 't Hof et al., "Evaluating The Effectiveness Of An E-Mental Health Intervention For People Living In Lebanon: Protocol For Two Randomized Controlled Trials."

⁶³ Interviewee 2.



Figure 9: Classification of current interventions according to pyramid level⁶⁴

⁶⁴ Authors' own production



IV. DIGITAL MHPSS: CHALLENGES AND SOLUTIONS

IV. Digital MHPSS: Challenges & Solutions

1. Safety

1.1 Data Protection

Challenge 1: Anonymity, Vulnerability and Data storage

Data protection and anonymity are of utmost importance, especially for vulnerable users. The information provided, for example, during online consultations entails highly **confidential** elements on personal life issues. Given the persistent stigma around mental health, this could severely harm service users if divulged to third parties.

Anonymity in digital interventions expands the access of MHPSS to populations who are particularly at **risk of stigma**. For example, LGBTQI+ populations in Syria that find it hard to seek face-to-face counselling, reach out for help thanks to the Fadfada Project developed by UNDP (an online psychosocial support platform providing confidential support through virtual media).⁶⁵ Vulnerable populations facing administrative penalties, such as undocumented migrants in the Netherlands, who face deportation if their situation is revealed, use a WhatsApp helpline to seek help.⁶⁶ These examples illustrate that breaches of anonymity can entail both psychological and physical safety risks.

Digital MHPSS interventions could be potential targets of **data theft and hacking**. Many NS do not have adequately **secured data storage facilities**. Remote services provision poses additional risks to data security. Where data security cannot be guaranteed, the potential harm to service users can be immense, as detailed above. Therefore, for example, the PEARL database deployed in Africa to store the medical files of patients was required by the ICRC MHPSS Regional Specialist to be hosted on a Geneva server.⁶⁷ However, as the breaches of ICRC servers illustrate, even higher protection capacities are needed to guarantee the safety of digital MHPSS interventions.



Solutions and Best Practices

Digital solutions need to guarantee minimal **data accessibility by design**. For example, the Step-by-Step web-based version tailors authorisation only to those who need it and for specific purposes only.⁶⁸

Moreover, services that ensure **end-to-end data encryption** should be favoured for online consultations, such as the Signal messenger app. Enhanced data protection can also be achieved by decentralised data storage on end-user devices or through blockchains, as compared to centralised servers.⁶⁹ However, security risks remain.

⁶⁵ Interviewee 4.

⁶⁶ Interviewee 5.

⁶⁷ Interviewee 14.

⁶⁸ Interviewee 1.

⁶⁹Margheri et al., "Decentralised Provenance For Healthcare Data."; Ramachandran et al., "Towards Complete Decentralised Verification Of Data With Confidentiality: Different Ways To Connect Solid Pods And Blockchain."

Challenge 2: Commercialisation of data.



To overcome technical challenges and finance gaps, NS might be inclined to partner with private businesses. However, this may in turn amplify safety concerns. Many companies who are actively engaged in digital humanitarian action like Facebook or Google, for example, rely on the **commercialisation of data** of the service users. Therefore, special precautions are warranted when cooperating with private companies.

Solutions and Best Practices

One way to overcome this concern is pioneered by the Danish Red Cross. In their cooperation with the private start-up 'Boblberg' to provide psychosocial support for people suffering loneliness, the Danish Red Cross employs a legally **innovative partnership** agreement that guarantees **co-ownership of the personal data** gathered thus impeding its commercialisation.⁷⁰



Boblberg⁷¹

Boblberg is a digital citizen-to-citizen platform to fight chronic loneliness developed by the Danish Red Cross in partnership with a private technology company.

The platform is available as a mobile app or website and functions as a simple message board where users can create a "bubble" (message) to make new friends, pursue hobbies, or share common experiences of life problems.

With over 400,000 users in Denmark, the intervention reached more than 7% of all Danes over the age of 15 and as many as 21% of all young people between the age of 15 and 24. In contrast to other social networks, there are no permanent profiles, safeguarding the anonymity of users to prevent negative effects of stigmatisation.

While the platform's code is owned by the partner organisation 'Boblberg', who also takes care of the day-to-day operations, the Danish Red Cross has employed an innovative partnership agreement securing the co-ownership of all user data generated to ensure non-commercialisation of data. The project is financed by charging the municipalities it operates in a fixed amount, which is sufficient to cover the operating, but entails significant costs for convincing and (re-) negotiating with the municipalities to extend the partnership.

For more information visit: https://boblberg.dk.

⁷⁰ Interviewee 8.

⁷¹ Ibid.

1.2 Do No Harm

Challenge 1: Unintended harm of interventions.



Digital MHPSS interventions may cause unintended harm to users when **misused**, or when digital MHPSS is employed in contexts where MHPSS is highly stigmatised. For example, the Mexican Red Cross offers crisis support to young people by using a specific hashtag on social media. This may, however, expose the service users to stigmatisation or abuse by publicly identifying their vulnerability.⁷²

Moreover, digital interventions need to be **updated** to remain safe. Otherwise, some features might become dysfunctional. Moreover, outdated information and could mislead users and lead to unintended harm in many possible ways.

Solutions and Best Practices

Organisations should only use **evidence-based interventions** to ensure safety.⁷³ Therefore, while organisations may want to develop a new digital MHPSS intervention tailored to their specific needs, such initiatives could be modelled based on **existing RCT-tested interventions** if they do not have the means to conduct an evaluation.

Protocols for all newly developed interventions should include **mandatory evaluation**, especially for specialised mental health care services, targeting more vulnerable users. For example, the digital version of the START Now project will be tested through an RCT to evaluate its safety for users before being scaled up and further disseminated.⁷⁴

Challenge 2: Dissemination of non-evidence-based digital interventions

Since many digital MHPSS technologies are still new and applied in a plethora of different contexts, the **evidence of many interventions is incomplete**. Evaluations tend to be complex and expensive, draining already scarce resources. Therefore, many interventions remain unassessed or are evaluated superficially. Private sector interventions, in particular, are incentivised to prioritise profit over safety.⁷⁵ However, interventions that are not evaluated and not evidence-based can pose severe harm to service users seeking help and should not be disseminated.



Solutions and Best Practices

To avoid the dissemination of non-evidence-based interventions, a 'whitelist' of rigorously evaluated interventions could be compiled and constantly updated, while considering factors of contextual and cultural adaptation. Conversely, non-evaluated interventions or such that have proven ineffective should be **blacklisted**. Funding organisations and donors should have access to such lists to make informed and safe choices for future service users.

⁷² Interviewee 22.

⁷³ Interviewee 29.

⁷⁴ Ibid.

⁷⁵ Interviewee 20.

1.3 Crisis intervention

Challenge: Limitations in case of acute crisis



Digital MHPSS interventions may fall short in delivering effective crisis care. While digital interventions allow for more flexible and immediate access to care, they also enable more abrupt break-off of contact, reducing the scope of **immediate intervention in acute crises**, compared to face-to-face consultations.

Solutions and Best Practices

To ensure patient well-being and prevent treatment setbacks, **digital technologies** should therefore only be employed with **caution** for the therapy of severe mental disorders.

It is advisable to **discuss reactions to crises** with the service user in advance and offer access to **updated digital emergency referral and service directories**, bearing in mind the costs this entails. Where no referral options exist, for example in remote contexts, further considerations of potential risks need to be carefully assessed before implementing digital interventions.

Moreover, digital MHPSS tools should feature an easily **accessible emergency button** ensuring a prompt response, notably in case of suicidality. Some web-based interventions, such as START Now, offer a digital 'panic' button to get immediate support from selected emergency contacts.⁷⁶

Finally, **clear standard operating procedures** should be established. These are necessary, e.g., to confirm the explicit consent of the user; and for high-risk cases (e.g., cases in which there is a risk of self-harm or identified protection needs) to ensure a prompt and adequate response by practitioners.

1.4 Practitioners' methods and well-being

Challenge 1: Misdiagnosis and medical error.



Practitioners' opinions on digital interventions such as teleconsultations are **split**. Some practitioners deem remote consultations equally safe and even prefer them for their flexibility and logistical simplicity, offering enhanced continuity of care as patients tend to miss fewer sessions.⁷⁷ Others, however, find themselves to be less service user-centred, more easily distracted in remote consultations, owing mostly to the lack of non-verbal communication. One informant reports an increased susceptibility to **potential misdiagnosis and malingering**.⁷⁸ This can lead to **poor quality of care and imprecise diagnosis** in the case of specialised care.

⁷⁶ Ibid.

⁷⁷ Interviewee 6.

⁷⁸ Interviewee 30.



Solutions and Best Practices

To avoid such issues, practitioners have emphasised the need to always **turn on the camera** in consultations, if possible. This enables the practitioner to be more patient-centred and assess elements of non-verbal communication.

Moreover, **professional training** is required to ensure practitioners are sensitive to the particularities of remote therapy. Volunteers and staff should also be sufficiently trained before implementing remote-guided interventions. Supervision should be put in place to allow practitioners' adaptation to the new method.

Finally, **hybrid models** appear most adequate to ensure diagnostical precision and facilitate rapport while simultaneously making use of the flexibility of remote delivery to optimise therapeutic continuity.

Challenge 2: Impediment on private life



While remote contact offers higher flexibility and continuity, it also brings constant **access** to practitioners even **outside office hours**. Therefore, practitioners might find it harder to separate duty and leisure and strike an adequate work-life balance.⁷⁹ Some practitioners might also be harassed during teleconsulting.⁸⁰



Solutions and Best Practices

Practitioners **shouldn't operate on personal devices** and shouldn't divulge personal contact information. Practitioners should be encouraged not to respond to contact requests outside of working hours. **Organisations** should therefore **provide the necessary devices** to staff and **set clear policies** on working hours.

To manage the increased pressure from constant accessibility, practitioners could benefit from **heightened peer support**, which can be delivered online for flexibility. This approach is taken, for instance, by the Netherlands Red Cross for its volunteers in response to heightened stress related to the COVID-19 pandemic.⁸¹ Moreover, practitioners must be supported in handling caseload, off-hour contacts, etc., through increased **supervision**. To avoid increasing time concerns, supervision should be scheduled during working hours and could be delivered remotely to allow more flexibility, e.g., via video call or messenger.

⁷⁹ Interviewee 27.

⁸⁰ Interviewee 16.

⁸¹ Interviewee 15.
2. Effectiveness

2.1 Positive impact



Challenge 1: Lack of exhaustive evidence

Even though an increasing number of research trials underline the effectiveness of digital MHPSS in specific contexts, the **evidence base is far from complete**. Most evidence on digital MHPSS relates to specialised mental health interventions such as online psychotherapy while there is a **gap in the literature** regarding the effectiveness of digital basic psychosocial support.

Furthermore, many trials are conducted in clinically controlled settings that have limited external validity to real-world circumstances.⁸² In addition, most trials were conducted in higher income countries, which may further limit their generalisability.⁸³ Thus, the use of **digital technology for MHPSS in limited-resource settings still requires substantial research.**⁸⁴ Finally, many digital interventions reviewed in this report are still being piloted and are yet to complete evaluation.⁸⁵

Solutions and Best Practices

A possible way to address the lack of exhaustive evidence and avoid the resource-intensive evaluations is to implement **already evaluated interventions** that have been proven effective. Moreover, NS should prioritise treatment approaches whose effectiveness in digital delivery is best-documented. This is the case notably for **interventions based on CBT**.⁸⁶ As mentioned above, a 'whitelist' of effective digital interventions could guide providers.

Challenge 2: Scepticism

A second challenge to ensure the positive impact of digital MHPSS relates to practitioners' scepticism towards the effectiveness of digital technologies. While some practitioners report successful treatment through digital technologies,⁸⁷ others feel the need to observe service users physically (non-verbal cues and behaviours) and do not think that digital interventions can substitute physical contact.⁸⁸ This in many cases pertains to personal preferences and professional styles, as well as scepticism on the safety of digital interventions.



Solutions and Best Practices

Practitioners should be able to flexibly choose the way they want to deliver help, keeping in mind that in certain situations such as lack of access, there might be no other options.

 ⁸² Ellis et al., "Implementation Of E-Mental Health For Depression And Anxiety: A Critical Scoping Review, 914.
 ⁸³Carter et al., "The Emergence Of Digital Mental Health In Low-Income And Middle-Income Countries: A Review Of Recent Advances And Implications For The Treatment And Prevention Of Mental Disorders", 224.
 ⁸⁴ Nachurd et al., "Digital Technology For Treatment And Prevention Of Mental Disorders", 224.

⁸⁴ Naslund et al., "Digital Technology For Treating And Preventing Mental Disorders In Low-Income And Middle-Income Countries: A Narrative Review Of The Literature.", 11.

 $^{^{85}}$ See Annex 2 – evaluation column.

⁸⁶ Attridge, "Internet-Based Cognitive-Behavioral Therapy For Employees With Anxiety, Depression, Social Phobia, Or Insomnia: Clinical And Work Outcomes" and Venkatesan et al., "Digital Cognitive Behavior Therapy Intervention For Depression And Anxiety: Retrospective Study".

⁸⁷ Interviewee 28.

⁸⁸ Interviewee 30.

2.2 Acceptance

Challenge: Lack of engagement



The sustained engagement of service users is a critical factor for effectiveness, which is even more challenging in digital interventions, especially those that are unguided.⁸⁹

Other factors affecting engagement include **credibility** in the evidence behind the content, as well as the **length** of the program – concise programs having higher rates of engagement - or the perceived **usefulness** of the intervention.⁹⁰ Concerning the latter, **expectations** about the intervention might be misdirected, as some users might expect a comprehensive individual therapy when they engage in interventions that do not deliver such.⁹¹



Solutions and Best Practices

A way to improve engagement is through the inclusion of **notifications** reminding users to regularly use the app/website.⁹² In addition, interventions should be designed to be as concise as the therapeutical base allows it, and **expectations** about what to expect from the intervention should be **managed in advance**.

Adequate and targeted promotion is a key factor for take-up and acceptance. For example, the Kenyan Red Cross partnered with the Ministry of Health to promote their MHPSS services by including the NS toll-free number in the Kenyan Ministry of Health awareness campaigns and social media, so that people know how to reach out for support.⁹³ Similarly, key actors in the community can played a crucial role in the promotion of the intervention.⁹⁴

2.3 Cultural appropriateness

One of the main problems when implementing an already-developed digital MHPSS intervention is that the original content might not be **socially/ culturally relevant** for the target population, which results in low engagement and acceptability. ⁹⁵ Culture affects the understanding of mental health and psychosocial wellbeing for everyone, and hence, it is essential to carefully consider it when designing and implementing a digital intervention.⁹⁶

Therefore, when implementing an intervention originally developed for a different target group, it is essential to **adapt it to the local context and target audience**. However, this process entails its own challenges.

⁸⁹ Interviewee 19.

⁹⁰ Borghouts et al., "Barriers To And Facilitators Of User Engagement With Digital Mental Health Interventions: Systematic Review.", 11.

⁹¹ Interviewee 16.

⁹² Interviewee 19.

⁹³ Interviewee 16.

⁹⁴ Interviewee 4.

⁹⁵ Borghouts et al., "Barriers To And Facilitators Of User Engagement With Digital Mental Health Interventions: Systematic Review.", 12.

⁹⁶ Interviewee 10.

Challenge 1: Adapting codes



Adapting existing self-guided interventions such as apps and websites requires alterations to the program's code, which might be protected by copyright or difficult to access.⁹⁷ Even where the code is accessible, **alterations require substantial technical expertise**. As the example of the Global Disaster Preparedness Center (GDPC)'s First Aid App illustrates, this can be a critical challenge, which can even inhibit the launch of interventions.

Adaption is a process that entails the accommodation of different preferences and interests – a balance needs to be achieved between what is affordable and easy to program, end-users' preferences, and the therapeutical core. ⁹⁸ During this process, and especially if the intervention requires substantial changes to be transposable, **the evidence base it rests on may become obsolete**. This would have important safety implications.



Solutions and Best Practices

To facilitate the process of adaptations, **open access codes** should be encouraged and adopted. For example, the WHO's intervention Step-by-Step is licensed under creative commons. Hence, NS could take the code and adapt it to their national contexts freely.⁹⁹ Moreover, the First Aid App was provided to NS to make any adaptions in visuals or content that they wished.¹⁰⁰

To realise the necessary technical adjustments, **technical assistance** to NS could be provided by the IFRC, or other partnerships, including private contractors.

In addition, if possible, adaptations should be made to the interface of a digital intervention (symbols, figures, etc.), **not the therapeutic substance.** Where more profound adaptions are necessary for acceptance, the intervention should be re-evaluated to **ensure compliance with safety standards.**

⁹⁷ Interviewee 9.

⁹⁸ Interviewee 31.

⁹⁹ The WHO would appreciate to be notified (Interviewee 2).

¹⁰⁰ Interviewee 1.

Universal App Program / First Aid App¹⁰¹

The First Aid App is a mobile application developed under the Universal App Program by the Global Disaster Preparedness Center, a cooperation between the IFRC and the American Red Cross. The Universal App Program is a "platform to facilitate the adaptation and localisation of mobile applications (apps) for use in countries across the globe."

The First Aid app is one of two mobile apps launched under this program and contains information on first aid. The app platform is centrally developed and can be used at no cost by NS who can adapt its content and design and take on ownership of the app on the national level.

Currently, the GDPC is planning a re-launch of the First Aid App which includes a section on Psychological First Aid. Due to lacking adaptions in the previous version which made the service unavailable in many countries, and because most adaptions were superficial rather than substantial, the new version is likely to launch globally using symbols and designs that are relevant for a global user base

For more information visit: https://preparecenter.org/activity/universal-app-program/

Challenge 2: Tailoring



Even if adaptions are more subtle, **tailoring them to the target group can be challenging**. For example, the Swiss and the Swedish Red Cross reported substantial difficulties in finding the correct terms and adequate tone for translations into Arabic. ¹⁰² Moreover, adequate promotion strategies need to be tailored to the specific context and target group.



Solutions and Best Practices

Firstly, to adapt a digital MHPSS intervention it is crucial to **clearly define the new target group** to narrow down the scope of adaptation.¹⁰³

Moreover, to ensure that the adaptation is culturally sensible, potential **service users should be involved in all the phases** of the intervention (see Section 5). Similarly, the **inclusion of native speakers and staff with profound cultural knowledge** of the intervention's context has the potential to greatly improve its acceptance and take-up.

Indeed, **even simple adjustments** such as the intervention's name may have an important effect. For example, UNDP Syria decided to call their intervention Fadfada (i.e., 'talking openly to someone') and avoid the term 'mental health', which is stigmatised in the local population. This example illustrates the importance of understanding the cultural context in which the intervention is going to be implemented.¹⁰⁴

¹⁰¹ Participant observation of GPDC First Aid App Roundtable, conducted February 14, 2022 on MS Teams.

¹⁰² Interviewee 7.

¹⁰³ Ibid.

¹⁰⁴ Interviewee 4.

Furthermore, it should be noted that **cultural adaptations can be facilitated by digital technology** if designed correctly. For example, being able to choose the features of the avatar, can easily make the intervention more culturally, and individually, appropriate.

To adapt digital interventions effectively, it is necessary to also consider **contextual adaption when selecting the digital technology** and platform through which the intervention is going to be implemented. For example, while in Syria WhatsApp is preferred, in some parts of Kenya the preferred technology is SMS.¹⁰⁵

Finally, it is worth considering that interventions need not be 100% perfect right away, as long as basic safety is guaranteed.¹⁰⁶ Echoing this, some practitioners encourage NS to pilot interventions **and make adaptions and improvements as-you-go**, based on trial and error, and user feedback.¹⁰⁷

Sui - Digital psychological and psychosocial support for refugees¹⁰⁴

The Swiss Red Cross in collaboration with multiple partners, including the Universities of Berne, Zurich, Lausanne, and FU Berlin, develops an internet-based guided self-help intervention called Sui for adult refugees living in Switzerland.

The multilingual digital information and intervention platform for mental health and psychosocial support is intended to sustainably improve the quality of life of refugees in Switzerland, by:

- restoring their sense of control and security and boosting their health literacy by providing psychoeducational information that is relevant to everyday life and offering details of support services. Concerns about family members left behind in conflict zones, an uncertain residential status in Switzerland and difficulties with social integration or not being recognized as a member of society are issues that particularly affect refugees.
- using exercises in emotional, problem and stress management to help reduce the impact of symptoms in their day-to-day life and prevent these symptoms from becoming chronic.
- boosting refugees' resources as well as their self-efficacy and decision-making ability.
- identifying mental stress and de-stigmatizing it, thus lowering the hurdles for requesting support.

The usage and efficacy of the digital information and intervention platform will be optimized with the support of trained helpers from the same target group at an individual time investment of 15 – 20 minutes per week.

The intervention will be evaluated from July 2022 until May 2023 in a RCT with refugees living in Switzerland.

The intervention runs on an open-source platform developed by the Swiss Red Cross and the Freie University Berlin. The platform will be released for adaptation and use in 2023.

For more information, contact: monia.aebersold@redcross.ch

¹⁰⁵ Engels, M, e-mail message to authors, March 19, 2022.

¹⁰⁶ Interviewee 1.

¹⁰⁷ Interviewees 8 and 19.

¹⁰⁸ Interviewee 7.

3. Scalability

3.1 Accessibility

Not all users can access all digital services equally, as some face structural, social, or individual barriers.

Challenge 1: Structural barriers

A primary concern is the **lack of internet coverage and/or end devices** for service users.¹⁰⁹ This amplifies concerns over increased digital poverty leading to social inequality and marginalisation for those with lower access, particularly in rural areas.





Solutions and Best Practices

Some technologies, such as mobile apps can be designed for **offline usage**. Thus, internet connection is required only upon download but not throughout usage. This solution has been adopted, for instance, by the Step-by-Step app.¹¹⁰

Another solution is to offer access to devices and stable internet in schools, community centres or religious institutions.

Challenge 2: Social and political barriers

(a) Social barriers



Mental health issues are still highly stigmatised. Some service users might not have sufficient **privacy** within their homes to confidentially access MHPSS services. This is true in many low-income contexts with overcrowded housing and is most acute in refugee camps.¹¹¹ For example, survivors of domestic violence are likely to find it difficult to use digital MHPSS within the vicinity of their families.



Solutions and Best Practices

Again, partnerships with schools, community centres or faith-based institutions could provide **safe spaces** for accessing MHPSS services. Moreover, **engaging community actors** might be beneficial for decreasing stigma in the long run.

(b) Political barriers

In some cases, where stigmatisation is politically institutionalised, these partnerships might not be viable. Some countries have severely repressive mental health legislation that dispossesses of basic rights those who seek help.¹¹²

¹⁰⁹ Interviewees 12, 26 and 32.

¹¹⁰ Interviewees 1 and 2.

¹¹¹ Interviewees 4 and 28.

¹¹² For example, under the Nigerian Lunacy Act which originated from colonial times, see Ugochukwu et al., "The Time Is Now: Reforming Nigeria's Outdated Mental Health Laws."



Solutions and Best Practices

In this case, NS must focus on **mental health advocacy**, alongside government and civil society partners to change and create appropriate rights-based mental health policy and legislation.

(c) Bureaucratic barriers

Even where stigmatisation is not institutionalised, there can be **bureaucratic hurdles** to scaling interventions. For specialised treatments, providers might need to acquire a medical certification which is very costly. Regulations to this effect differ strongly internationally, entailing practical challenges for transposing interventions to different contexts. The same can be true even between provinces regions within one country, as is the case for Canada.¹¹³ Moreover, in Lebanon, for example, digital medical services are generally not licensable.¹¹⁴



Solutions and Best Practices

Considering this, interventions on the lower bounds of the pyramid that require less medical proficiency are likely to have higher scalability.

Challenge 3: Individual barriers



A further barrier concerns **literacy**. Children, marginalised groups, and people in certain lowincome countries are particularly affected by this impediment to access care. The issue is less pressing for interventions with personal contact like phone or video hotlines, but more limiting for automatised interventions such as mobile apps. There is also the particular concern of lacking '**digital literacy'** often faced by older persons.¹¹⁵



Solutions and Best Practices

Illiteracy can be overcome by prioritising **audio and video content** within the intervention. For example, the Step-by-Step, ¹¹⁶ or Sui mobile applications ¹¹⁷ integrate therapeutic elements with fictional narratives and the use of visuals and audio. **Digital illiteracy** can be remedied by designing simple and intuitive user interfaces, providing IT guidance services, or integrating 'tutorials' on how to use the intervention. The issue is also likely to be less acute for guided interventions.

¹¹³ Interviewee 18.

¹¹⁴ Interviewee 4.

¹¹⁵ Interviewee 31.

¹¹⁶ Interviewee 2.

¹¹⁷ Interviewee 7.

3.2 Resource limitations



Challenge: Lack of funding, expertise, and time

The main challenge that is shared by virtually all NS concerns the **inadequate funding** for technical solutions, in-house expertise to administrate them, and suitably **specialised staff** to design adequate interventions. Digital interventions require high investments not only in software development, but also data storage hardware, and legal expertise to ensure compliance with dynamically changing data protection regulations. Moreover, developing reasonably complex, safe, and adapted digital interventions requires much **time**; one informant estimates an average of three years from inception to launch.¹¹⁸ Additionally, they require long-term funds for promotion, maintenance, and enhancements.¹¹⁹



Solutions and Best Practices:

(a) Adapt centralised apps

Instead of 'reinventing the wheel', NS should 'build from what we have', ¹²⁰ by **using existing platforms**. For example, the WHO offers the 'Step-by-Step' mobile app, a software filed under creative commons license which can be used and adapted by any organisation.¹²¹ Similarly, the GDPC offers two mobile applications under their 'Universal App Program' that can be used and adapted by NS at '**minimal cost'**.¹²² The Swiss Red Cross provides a similar service through its technical platform in cooperation with the Freie University Berlin. To minimise the running costs for maintenance and enhancements of interventions based on these universal platforms, a Digital System that is centrally updatable could be introduced, for instance by the IFRC.

(b) Partnerships

There are several well-established partnership models in the Movement that facilitate costefficient knowledge and technology sharing, which can be leveraged to spread digital MHPSS.

The IFRC in general offers a variety of resources from psychoeducational learning materials, capacity-building workshops (e.g., MHPSS in emergencies, or PFA), staff and volunteer training modules (e.g., on supervision or suicide prevention), to operational guidelines on various topics related to MHPSS. Platforms promoting **collaboration between NS** should also be leveraged. Scaling up interventions that have proven safe and effective for one NS

¹¹⁸ Interviewee 1.

¹¹⁹ Interviewee 9.

¹²⁰ Interviewee 2.

¹²¹ Ibid.

¹²² Interviewee 11.

to another context saves resources, especially in digital technologies which are easy to replicate by nature.

Public partnerships can also offer viable solutions. The Zambian Red Cross successfully streamlined MHPSS in the national response to the COVID-19 pandemic by partnering with the Zambian Ministry of Health, benefitting from collaboration with The Netherlands Red Cross.¹²³

Where expertise for adaptions is needed or where building a new system is necessary, **collaborations with the private sector** may be justified. The Danish Red Cross, for instance, has minimised its technical expenses by partnering with a private company named 'Boblberg' to provide a highly successful psychosocial support intervention for people suffering loneliness.

Experience from the private sector, moreover, suggests that even complex mobile apps can be commissioned at reasonable costs given the growing international market for such services.¹²⁴

(c) Low-tech interventions

In most contexts, costs may be reduced by implementing simpler technical interventions, employing technologies that are already integrated into the daily workflow of NS.

Phone hotlines by NS have been in operation for various purposes for several decades. More recently, **messenger apps** are being used for consultations and supervision, e.g., by the Zambian Red Cross for COVID-19 health workers and volunteers.¹²⁵ **Video calls** can be used for the same purposes, as well as conducting training in, e.g., Problem Management Plus (among others by the IFRC Reference Centre for Psychosocial Support), Self Help Plus, or Psychological First Aid. **Social media** is routinely used for communication, fundraising, advocacy, promotion, etc., and can equally be leveraged to increase awareness about mental health issues and even implement MHPSS.

¹²³ Interviewee 32.

¹²⁴ The App Bipolar Buddy that employs AI technology to predict mood swings of people with bipolar disorder was developed for a total of \$30,000 by a specialised technology company from Bangalore, India, (Interviewee 5). ¹²⁵ Interviewee 32.

4. Sustainability

4.1 Temporal Sustainability

Challenge: Limited Financial and Human Resources



Digital MHPSS may entail **high long-term costs** for maintenance and enhancement. Interventions that are app- or web-based require constant updates to improve user experience, warrant data protection, update information (e.g., local service directories),¹²⁶ maintain compatibility with IT software and hardware (e.g., iOS compatibility for apps), and comply with changing regulations.¹²⁷ All of these require long-term and continuous financing, which is one of the main challenges, according to many informants. Without long-term funding, digital MHPSS services risk being discontinued.¹²⁸

Maintenance and updates also require **substantial staff time**. For roles that require less specialised and professional knowledge and are performed by volunteers, the instability of their contribution can negatively impact service delivery.¹²⁹ Where inputs by specialised staff such as psychologists and psychiatrists are required, the lack of long-term funding could jeopardise their presence, and with it, the safe and effective implementation and evaluation of interventions. For example, a Virtual Reality intervention against chronic pain used in several ICRC physical rehabilitation centres in South Asia was planned and delivered, but in one country, the project was suspended due to the removal of a psychologist from the team.¹³⁰



Solutions and Best Practices

To increase sources of funding, practitioners could reach out to various potential funding sources outside of the Movement, such as **philanthropic endowment** funds by the private sector,¹³¹ donations of **faith-based organisations**.¹³² Considering the high costs of rigorous evaluations in money, staff, knowledge and time, **academia** could also be an important partner in evaluating digital MHPSS interventions and engaging in continuous monitoring and improvements and integrating automated evaluations in digital interventions. For example, the Kenyan Red Cross engaged the private sector for funding to support its digital MHPSS initiatives,¹³³ and the Step-by-Step app is funded by the European Union.¹³⁴

¹²⁶ Participant observation of GDPC First Aid App Roundtable, conducted February 14, 2022 on MS Teams.

¹²⁷ Interviewee 9.

¹²⁸ Interviewee 1.

¹²⁹ Interviewee 13.

¹³⁰ Ibid.

¹³¹ Interviewee 29.

¹³² Interviewee 10.

¹³³ Interviewee 16.

¹³⁴ Interviewee 1.

4.2 Structural Sustainability

Challenge: Limited Embeddedness



A systematic review of 14 studies evidenced the **difficulty of institutionalising** new MHPSS innovations.¹³⁵ Undoubtedly, the structural sustainability of digital MHPSS hinges on the existing IT and mental health infrastructure they are embedded in. Where levels of IT infrastructure, knowledge/awareness for MHPSS, and availability of MHPSS personnel are reported low, ¹³⁶ there were considerable difficulties in delivering structurally sustainable digital MHPSS initiatives.



Solutions and Best Practices

(a) Regions with lesser-developed IT and MHPSS Infrastructures

Digital MHPSS can be leveraged to **increase awareness** for MHPSS, **popularising and destigmatising MHPSS**, which is a key step to making MHPSS more of a local priority and encouraging resources to be devoted to enhancing MHPSS infrastructure.¹³⁷ Digital MHPSS solutions with low investment requirements, such as information sharing on social media sites, are well-suited for this purpose.

(b) Regions with better-developed IT and MHPSS Infrastructures

Digital MHPSS can be more structurally sustainable by **including local healthcare information** and **establishing a referral system** to more specialised care if needed. For example, the Step-by-Step app includes local healthcare information specific to countries where refugees are located.¹³⁸ Practitioners can also consider developing evidence-based digital MHPSS into **medical devices**, such as with functions of tailoring treatment programmes to symptom profiles, thereby increasing the chance of being **supported by insurance schemes**. This is instrumental for improving both the temporal and structural sustainability of digital MHPSS.¹³⁹

Moreover, digital MHPSS would benefit from establishing **linkages with the national health system, existing governmental and professional bodies**. Below are two exemplars from the Argentinian and Canadian Red Cross respectively:

 ¹³⁵ Troup et al., "Barriers And Facilitators For Scaling Up Mental Health And Psychosocial Support Interventions In
 Low- And Middle-Income Countries For Populations Affected By Humanitarian Crises: A Systematic Review."
 ¹³⁶ Interviewees 22, 25, 26 and 32.

¹³⁷ For example in Lebanon, Zambia, Mexico, and some Pacific nations, see Interviews 4 and 26.

¹³⁸ Interviewee 1.

¹³⁹ Interviewee 29.

¹⁴⁰ Paul et al., "Increasing Psychological Service Access For COVID-19-Impacted Frontline Workers: Canadian Red Cross & Ontario Psychological Association Partnership."

¹⁴¹ Interviewee 18.

Remote Assistance Service for the Elderly¹⁴²

The Argentinian Red Cross has implemented a remote assistance service that help the elderly with emergency management or accompany the elderly when they need someone to talk to, in order to prevent onset of mental health issues.

The service user has been previously referred to the programme by their general practitioners (GP)/ other services from the national health system. On one hand, this digital intervention can fulfil the national healthcare goal of assisting vulnerable old people. On the other, it can also target particularly old people who have some level of need as identified by their GP.



E-referral System during COVID-19143

The Public Health Agency in Canada and Ontario Psychological Association (OPA) possess a digital system that registers people who need top-tiered specialised mental health support. The Canadian Red Cross leveraged this system, to match individuals who are quarantining with psychologists that suit them (e.g. based on what licenses the psychologists possess).

Initially it started in Ontario, but it spread across urban centres in the nation. More than half of the respondents waited no more than two weeks before their first session with a psychologist.¹⁴⁴ The initiative is structurally sustainable – The Canadian Red Cross managed intake and referrals while OPA provided pro bono access to MHPSS, while the partnership was embedded the infrastructures of the federal government.

Taken together, structurally sustainable digital MHPSS interventions should be integrated into national strategies and structures holistically to build synergy with the broader health system and interventions. This would enable higher operational effectiveness in delivering MHPSS services (such as alleviating the workload of practitioners and targeting people who need the service), more holistic care of service users, and eventually strengthening local IT and MHPSS capacities.

¹⁴² Interviewee 31.

¹⁴³ Interviewee 18.

¹⁴⁴ Paul et al., "Increasing Psychological Service Access For COVID-19-Impacted Frontline Workers: Canadian Red Cross & Ontario Psychological Association Partnership."

5. Participation

Challenge 1: Retaining users



With less monitoring on the use of and compliance to the intervention, there tends to be a **higher drop-out rate** in self-guided interventions (as compared to MHPSS with more nonvirtual elements such as in-person interaction).¹⁴⁵ For example, the Kenyan Red Cross reported that users who seek help via chatbots routinely disengage in mid-process.¹⁴⁶ Unsurprisingly, it is also more difficult to retain users to participate in evaluation processes.¹⁴⁷



Solutions and Best Practices

(a) Human-centred and participatory approach in protocols

Despite often limited financial and human resources, it would be helpful for practitioners to prioritise a **human-centred and participatory approach in their protocol**.¹⁴⁸ Service users should be involved from consultation, planning, to implementation, and eventually monitoring and evaluation. There are many **toolkits** that guide effective inclusion of service users (e.g., by UNHCR)¹⁴⁹ and adopting a human-centred design (such as that developed by IDEO).¹⁵⁰ Practitioners can integrate and adapt these toolkits to facilitate bottom-up co-creation of interventions that respond and align to users' needs better.

(b) Skilful engagement of target group

In the development of the Sui app, the research team conducted multiple **focus groups** to seek feedback on content and translation.¹⁵¹ Shah advocates for the approach of **'active invitation of pushback'** where practitioners encourage criticisms and honest feedback.¹⁵² It heavily predicates on creating a safe and welcoming environment, where, for example, consultants engaging with stakeholders and key informants are not defensive, and power dynamics cutting across race, gender, age etc are taken into consideration.¹⁵³

 $^{^{\}rm 145}$ Interviewee 1.

¹⁴⁶ Interviewee 16.

¹⁴⁷ Interviewee 19.

¹⁴⁸ Interviewee 3.

¹⁴⁹ United Nations High Commissioner for Refugees, "Effective inclusion of refugees Participatory approaches for practitioners at the local level."

¹⁵⁰ IDEO, "Design Kit: The Human-Centered Design Toolkit."

¹⁵¹ Interviewee 7.

¹⁵² Shah, "Ethical Standards For Transnational Mental Health And Psychosocial Support (MHPSS): Do No Harm, Preventing Cross-Cultural Errors And Inviting Pushback."

¹⁵³ One possible way to achieve this is that practitioners could propose an imaginary situation where someone has done something adverse for the community, ask community members what appropriate ways are to channel their concern and adapt those methods in the active invitation of pushback

(c) Representation in the development team

Providers might benefit from not only engaging user groups in participatory exercises but **including them as key team members** in every step of development and implementation.. For example, in Sui by the Swiss Red Cross, a Syrian team member oversaw cultural adaptation and translation processes. In the Step-by-Step app for Syrian refugees, a Syrian colleague was involved in the conception of the project.¹⁵⁴ Undoubtedly, one person cannot represent the entire target population, but given the difficulty of engaging user groups throughout, it might be a reasonable compromise. **Increasing diversity** within teams would certainly help mitigate biases and improve outcome

Below, we summarise the findings of this chapter in a comprehensive table and synthesise key solutions into a practical process chart to guide stakeholders throughout the process of launching digital MHPSS interventions.

Facet	Challenges	Solutions	Addressee	Examples
Safety				
Data protoc	Anonymity	Restricted access to data for some actors	NS	Step-by-Step
tion	Data storage	End to end data encryption	NS	Signal
FacetSafetyData protec- tionDo no harmDo no harmCrisis interventionPractition- ers' meth- ods and wellbeingEffectivenessPositive impactAcceptanceCultural ap- propriate- ness	Commercialisation	Co-ownership of data gathered	NS	Boblberg
	Unintended harm of in	Evaluate new interventions through RCTs	NS	START Now
Do no harm	tonintended harm of in-	Avoid the dissemination of non-evidence-based		
	terventions	interventions using a 'whitelist'	IFAC F3C	
Crisis	Delivering origin core	Easily accessible emergency button	AddresseeEXaNSStepNSSigrNSBobNSSTAasedIFRC PSCIFRC PSCIFRC PSCPractitionersIFRC PSCNSPractitionersIFRC PSCNSNS/IFRC PSCNetNS/IFRC PSCNetNS/IFRC PSCNetNS/IFRC PSCNetNSNSNSNSNSNSNSNSNSStepNS	START Now
intervention	ChallengesAnonymityData storageCommercialisationUnintended harm of in- terventionsDelivering crisis careMisdiagnosis and medi- cal errorImpediment on private lifeLack of exhaustive evi- denceScepticismLack of engagementAdapting correctly	Develop SOPs on high-risk cases	IFRC PSC	
	Misdiagnosis and modi	Use the camera in consultations	AddresseeExamor some actorsNSStep-NSSignaeredNSBoblkthrough RCTsNSSTARnon-evidence-based elist'IFRC PSCy buttonNSSTARcasesIFRC PSCItionsPractitionerscasesIFRC PSCtionsPractitionersonlyPractitionersrsNS/IFRC PSCNSNS/IFRC PSCed interventionsNSe effectiveness in (e.g. CBT therapy)NSrid approaches when sNSectationsNSenditionNSstance when adaptingNSstance when adaptingNSpopulation in all the cationsNSNSSwissnologyNSdifferenceNSSAHA	
Facet C Safety A Data protection A Do no harm U Crisis D intervention M Practition- M ers' meth- M ods and Ir wellbeing Ir Brositive Ir Acceptance La Acceptance A Cultural ap- propriate- ness A	ivilsulagnosis and medi-	Training	IFRC PSC	
	carerror	Hybrid models	NS Descritizione en	
ods and	Impediment on private	Use non-personal devices only	Practitioners	
wellbeing	life	Increase peer support offers	NS/IFRC PSC	Netherlands RC
	Supervision		NS/IFRC PSC	Zambian RC
Effectiveness				
Positivo	Lack of exhaustive evi-	Implement already evaluated interventions	NS	
		Prioritise approaches whose effectiveness in	NS	
impact	dence	digital is well documented (e.g. CBT therapy)	145	
impact	Scenticism	Promote flexibility and hybrid approaches when	NS	
	Scepticisti	possible	Addressee Examples NS Step-by-Si NS Signal NS Boblberg RCTs NS START No ence-based IFRC PSC NS START NO IFRC PSC Practitioners IFRC PSC NS Practitioners NS/IFRC PSC NS Practitioners NS/IFRC PSC NS Practitioners NS/IFRC PSC NS NS eness in therapy) aches when NS NS NS NS NS NS NS NS NS NS	
		Include notifications and reminders in the case	AddresseeExamplesta for some actorsNSStep-by-SttionNSSignalatheredNSBoblbergions through RCTsNSSTART Nonn of non-evidence-based thitelist'IFRC PSCency buttonNSSTART Nonisk casesIFRC PSCultationsPractitionersultationsPractitionersoffersNS/IFRC PSCNSNS/IFRC PSCuluated interventionsNSthose effectiveness in ted (e.g. CBT therapy)NShybrid approaches when ionsNSvespectationsNSexpectationsNSsubstance when adapting ionNSsubstance when adapting ionNSswiss RC ionSwiss RCionNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNSsubstanceNS<	
Accentance	Lack of engagement	of automated interventions		
Acceptance	Lack of engagement	Manage service users' expectations	NS	
		Adequate and targeted promotion	iss to data for some actors NS Step-by-Step a encryption NS Signal of data gathered NS Bobiberg nterventions through RCTs NS START Now mination of non-evidence-based ising a 'whitelist' e emergency button NS START Now on high-risk cases IFRC PSC a in consultations Practitioners IFRC PSC NS nal devices only Practitioners support offers NS/IFRC PSC Netherlands RC NS/IFRC PSC Zambian RC 2000 2000 2000 2000 2000 2000 2000 20	Kenyan RC
	Adapting codes	Instance actions to reactionINSEnd to end data encryptionNSCo-ownership of data gatheredNSEvaluate new interventions through RCTsNSAvoid the dissemination of non-evidence-based interventions using a 'whitelist'IFRC PSCEasily accessible emergency buttonNSDevelop SOPs on high-risk casesIFRC PSCUse the camera in consultationsPractitionTrainingIFRC PSCHybrid modelsNSUse non-personal devices onlyPractitionIncrease peer support offersNS/IFRCSupervisionNS/IFRCSupervisionNSPrioritise approaches whose effectiveness in digital is well documented (e.g. CBT therapy)NSPromote flexibility and hybrid approaches when possibleNSInclude notifications and reminders in the case of automated interventionsNSAdequate and targeted promotionNSUse open-source codesNSProvide technical assistance to NSIFRC PSCDon't alter therapeutic substance when adapting phases of the interventionNSAdapt to the preferred technologySimple adaptations make a differenceNSPiloting & continuous adaptationsNS	NS	Step-by-step
	Adapting codes	Provide technical assistance to NS	IFRC PSC	
Cultural ap-		Don't alter therapeutic substance when adapting	NS	
propriate-		Involvement of the target population in all the	NS	Swiss BC
ness		phases of the intervention	145	5W155 NC
	Adapting correctly	Adapt to the preferred technology		
		Simple adaptations make a difference	NS	Fadfada
		Piloting & continuous adaptations	NS	SAHA

¹⁵⁴ Interviewee 1.

Scalability					
	Lack of end devices /	Design tools that	NS	Step-by-Step	
Accessibility	stable connection	Partner with com	munity actors to provide safe	NS	
	Insufficient privacy	spaces with stable	e connection		
	Bureaucratic hurdles	Prioritise low-leve	el digital tools for scale-up	NS	
		Adapt existing int	erventions	NS	Stan_by_Stan
		Centralised Digita	al Platforms	IFRC PSC	First Aid App
	Lack of funding	Leverage IFRC PS	C Resources	NS	Thist Ald App
Resource	Lack of IT expertise	Cooperate with o	ther NS	NS/IFRC	Zambian RC
Limitations	Lack of psychology exp.	Partner with nation	onal governments	NS	Zambian RC
	Lack of time	Partner with priva	ate contractors ensuring con-	NS	Boblberg (Danish
		tractual privacy a	nd safety	145	RC)
		Employ low-tech	interventions	NS	#NUJA (Mex. RC)
Sustainability					
	Limited financial and	Partnerships for f	unding and human resources		
Temporal		(e.g., philanthrop	ic endowments, faith-based	NS	Kenyan RC
	numanresources	organisations, go	vernmental bodies, academia)		
	Embed in existing infra- structures	Lesser-devel-			
		oped IT and	Use digital technology to	NS	
		MHPSS infra-	popularise MHPSS		
		structures:			
			Include local health service	NS	Step-by-Step Canadian RC
Structural		Higher-devel-	information		
		oped IT and	Establish a referral system	NS	Canadian RC
		MHPSS infra-	Inclusion in insurances	NS	
		structures:	Integrate with national		Argentinian RC
			health system, gov. & pro-	NS	0
			fessional bodies		
Participation					
	Retaining participants	Adopt a human-c	entred design toolkit	NS	Toolkit devel- oped by IDEO
		Prioritise particip vitation of pushba	atory approach and 'active in- ack' in protocols	NS	Sui App
		Engaging the targ ment team	et group within the develop-	NS	Step-by-Step

Table 2: Summary Table of common challenges and viable solutions

Illustration: Best Practices in Digital MHPSS

Design



Figure 10: Flowchart Illustration of Best Practices in Digital MHPSS¹⁵⁵

¹⁵⁵ Authors' own production.



V. RECOMMENDATIONS



V. Recommendations

1. To the Movement: Collect and leverage data

To ensure safety, improve service delivery and contribute to the frontier of knowledge in digital MHPSS, NS should increase data collection and evaluation of their interventions, in partnership with universities. This can include quantitative analyses, as well as qualitative data from interviews, focus groups or key informant interviews.

Where possible, providers should consider integrating **automated monitoring and evaluation** into digital interventions, in partnership with academia. This could include:

- Mood tracking of service users throughout time
- Completion of online questionnaires (such as PHQ-9) throughout time
- Big data analysis of user meta-data

Data gathered should be leveraged not only for scientific research but also for **needs assessment and planning**. Underscoring the potential of both big data and hybrid models, the 'Boblberg' tool, for instance, used big data analysis to identify a geographical cluster of single mothers connecting via their digital psychosocial support platform to launch an in-person family support intervention in their area.

2. To the Movement: Favour hybrid models

Scientific evidence warrants **realistic optimism** with regards to the benefits of digital MHPSS. Nonetheless, given its resource-intensiveness, as well as specific limitations in safety and available evidence, practitioners should always **critically question whether digital delivery is necessary** or whether conventional MHPSS could be equally beneficial. Where reasonable optimism is met by critical assessment, **hybrid delivery models which combine digital and conventional MHPSS modules will often emerge as the optimal option**.

a. Hybrid delivery of single interventions

If carefully designed, neither format must supersede the other and both can reinforce each other within a same intervention.

Digita	l technology can complement face-to-face	Personal contact can complement digital			
interve	entions through:	interventions through:			
-	Combining face-to-face sessions with flexible on- demand and just-in-time modules that can be followed at individual pace. Complimentary usage while awaiting treatment or in-between session Flexible delivery of complementary activities, such as coordination, training, and supervision.		Building a trustful and empathetic rapport Analysing behaviours and non-verbal commu- nications Reassuring and guiding service users Enabling enhanced crises intervention		

b. A hybrid model for integrated MHPSS programmes

Hybrid models are particularly valuable in comprehensive MHPSS programmes that link different interventions across the MHPSS pyramid. Based on the available evidence, we propose a tentative hybrid MHPSS pyramid with varying digital inputs at each level, linked to each other via a digital referral system.

The high scalability of digital interventions is particularly useful for basic psychosocial interventions that target large audiences. Here, self-guided interventions such as mobile/web apps appear the most promising. It should, however, be noted that the evidence for digital interventions at this level is incomplete. Given the relatively lower level of vulnerability of users, this mustn't bar implementation, but warrants piloting interventions with heightened monitoring and safety standards, especially for data security and anonymity.

Focused psychosocial support for specific groups at risk can benefit from a range of digital tools. The more successful ones are likely to be remotely guided, e.g., remote training (PM+), or targeted hotline services via messenger, phone, or video call.

NS may consider implementing digital psychological support interventions, best complemented through online or in-person guidance by trained implementers. In particular, NS can draw on a growing body of CBT-based mobile/web apps whose effectiveness and safety are increasingly well established. However, they should strictly monitor and evaluate the intervention as most evidence is based on clinical trials rather than real life emergencies.

Given maximum safety standards for particularly vulnerable service users, face-to-face delivery by specialists should remain the standard for specialised mental health care. It could be complemented by digital tools such as mood tracking or gamified exercises awaiting treatment or in-between sessions. Remote therapy is possible, after having established a personal rapport, in accordance with the preferences of practitioners and service users.



Figure 11: A Hybrid Model for integrated MHPSS programmes

3. To the Movement: Adapt available digital tools

Developing reasonably complex, safe, and adapted digital interventions requires substantial staff and financial commitments and is estimated to take approximately three years from inception to launch. To overcome resource gaps and ensure safety, NS should instead make use of the unique advantage of digital tools by replicating available digital tools. The following recommendations should be observed:

First, the IFRC should map existing digital interventions:

- a. A "whitelist" of safe and effective interventions should be established,
 - prioritising open-source models, e.g, Step-by-Step, PM+, SH+ from the WHO and SUI from the Swiss Red Cross
 - drawing on the robust evidence base for digital CBT-based interventions in particular
- b. A "**blacklist**" of unreliable interventions should be established in coordination with academia, to identify non-evidence-based interventions that should not be disseminated
- c. Both lists should be
 - disseminated amongst donors for reference on which projects to support
 - disseminated to national authorities to improve regulation of the private sector
 - regularly updated

Second, NS should focus available resources on making necessary **adaptations** to local culture and context to warrant take-up and effectiveness of the interventions, using **human-centred and participatory approaches**.

Third, the IFRC should establish a **Movement-wide MHPSS innovation centre** that centrally develops, maintains, updates, and evaluates new open-source digital interventions that are warranted by international needs. This will **minimise marginal costs** for NS. A relevant example is the Technical Platform developed by the Swiss Red Cross in cooperation with the Freie Universität Berlin, or the GDPC's Universal App Program which already developed two custom-made and adaptable IRCRCM apps, namely the "Hazard App" and the "First Aid App". The latter is currently being redesigned to include PFA.

Again, adaptions to local culture and context that increase take-up and effectiveness of the intervention are key.

4. To the Movement: Engage in principled partnerships

NS should leverage partnerships to benefit from the diverse expertise of, and synergies with, different entities in digital MHPSS. Partnerships may include working hand in hand with:

Entity	Objective
Universities	To develop and evaluate projects and build upon existing research
Local community actors	To provide safe spaces for access to internet-based interventions.
	E.g., schools, faith-based organisations, municipalities, or community centres
	To facilitate participatory, inclusive design and acceptance from user groups
Philanthropic organisations	To mobilise financial resources and increase the visibility of digital MHPSS
Governmental agencies	To raise awareness about the accessibility of MHPSS digital tools, ensure inte-
	gration into the broader health care system and ensure equal access
Private companies	To perform complex technological tasks and benefit from knowledge-transfers

Since private companies have a competitive edge in technological solutions, increased reliance on digital solutions will entail a certain dependence on private companies. The profit maximisation imperative of private actors may, however, collide with humanitarian principles. For instance, private companies might engage in partnerships for conscious laundering purposes or to gain access to the data of vulnerable populations.¹⁵⁶

Therefore, Business-Humanitarian Partnerships should be critically assessed.¹⁵⁷ According to IRCRCM guidelines, the following needs to be taken into consideration:¹⁵⁸

- Private companies should assist the humanitarian organisation in fulfilling its mission and not its own business interests. Particular caution should be applied with regards to noncommercialisation of data.
- The business partner must not be engaged in activities running counter to the Movement's objectives and principles and should not endanger its neutrality and independence.
- Companies that could compromise the reputation, neutrality, and impartiality of the Movement, e.g., by being materially involved in armament, health-endangering activities, or contributing to humanitarian emergencies are unfit for partnerships.¹⁵⁹

Bearing in mind the inherent power-asymmetry between humanitarians and populations in need, for instance concerning the issue of 'informed consent', partnership agreements need to observe at least the following elements:

- Non-commercialisation of services and user data
- Data security
- Prohibition of implementing non-evidence-based interventions

5. To the Movement: Build synergies with local structures

Digital MHPSS interventions should be integrated holistically into national strategies and structures to **build synergies with the broader health system**. This would allow:

- The task transfer within MHPSS between the Movement and the broader health structures
- Enhanced referral options to specialised services
- Accessing more data for more accurate needs assessments
- Higher reach for mental health advocacy and promotion

To maximise sustainability, interventions should also aim **at building local capacities**:

- Experts, preferably from the local community, should be included initially to build the capacities of local volunteers through a 'Train the Trainer' format (e.g., in PM+ or PFA)
- Implementers could guide end-users of MHPSS digital interventions, enhancing local IT literacy
- Support groups between volunteers should be developed as a network to transfer knowledge

Finally, where the institutional environment creates barrier for (digital) MHPSS, the Movement should partner with local stakeholders to **advocate for de-stigmatisation and enhanced legislation**.

¹⁵⁶ Burns, "New Frontiers Of Philanthro-Capitalism: Digital Technologies And Humanitarianism."

¹⁵⁷ Carbonnier and Lightfoot, "Business In Humanitarian Crises For Better Or For Worse?"

¹⁵⁸ Council of Delegates of the International Red Cross and Red Crescent Movement, *International Review of the Red Cross.*

¹⁵⁹ Ibid.

6. To the IFRC: Increase offers in digital training, SOPs, and supervision for practitioners

The digitalisation of services is an era-defining trend that will only become more relevant as the digital infrastructure continues to grow. This warrants swift and extensive organisational investments, not only in budgets and IT systems, but also staff capacities.

To enhance the quality of care and the wellbeing of staff and volunteers, the IFRC PS Centre should **develop SOPs** that guide practitioners engaged in digital interventions, and train and supervise them in their implementation.

The SOPs should cover aspects of telediagnosis, methodological advice (e.g., leaving the camera on to observe non-verbal communication), technical assistance (user guidelines for relevant tools), and practitioner's wellbeing (e.g., no contact outside of office hours). Specific SOPs should furthermore be developed to improve remote crisis intervention. To ensure effectiveness, the guidelines should leave room for adaptation to different cultural contexts and individual preferences.

The IFRC PS Centre should moreover **offer training to diffuse the SOPs, and increase supervision** for practitioners to ensure correct implementation and sustainable wellbeing. Digital technologies should be harnessed to efficiently scale up these services.



VI. CRITICAL Considerations



VI. Critical Considerations

1. 'WEIRD bias' in psychology publications

There are systematic biases in psychological studies from excessively relying on samples from Western, Educated, Industrialised, Rich, and Democratic (WEIRD) societies. ¹⁶⁰ This implicitly assumes that participants from WEIRD countries represent the 'standard' or 'norm', creating post-colonial epistemological biases, omitting the fact that norms of human behaviour and perceptions differ substantially across cultures. Thus, the external validity of the evidence base on which (digital) MHPSS interventions rest must be critically reviewed.

2. Inequitable access to digital MHPSS

Currently, the Global North is generally more well-resourced for provision of (digital) MHPSS than the Global South. The rate at and extent to which Global North and South benefit from digitalisation is not uniform and might thus reinforce and exacerbate existing inequities.

Moreover, the gender digital divide affects the accessibility of women to digital devices in comparison to men. Women, especially in developing countries might be less able to engage with digital MHPSS solutions than men.¹⁶¹ Furthermore, MHPSS needs are affected by biology, and socio-cultural norms around gender. Men, for example, seek psychological help less while women are more likely to suffer from SGBV. Beyond binary definitions of gender, intersectional oppression, and marginalisation along lines of race, sexual orientation and nationality create vulnerabilities that must be considered in targeting interventions. Practitioners should strive for more equitable access in their design and implementation of digital MHPSS.

3. Enabling environment for the holistic wellbeing of individuals and community

MHPSS is just one puzzle pieces for advancing the wellbeing of individuals and communities and should be complemented by other health support interventions. For example, practitioners can use digital MHPSS to direct users to different resources suitable for their mental health needs, such as medicine and government facilities. However, practitioners and users alike reach a dead-end when medicine is unaffordable, or when transport cost to government facilities is high.¹⁶² Moreover, apart from health concerns, it is also crucial to address root causes of vulnerabilities,¹⁶³ such as unemployment, poverty, and disruption of social networks. Digital MHPSS is meant to address specific needs but many more aspects are vital to the holistic wellbeing of individuals and communities.¹⁶⁴ Digital MHPSS should be used in conjunction to structural measures that aim at improving overall quality of life, such as legal and economic security.

¹⁶⁰ Henrich, Heine and Norenzayan, "The Weirdest People In The World?."

¹⁶¹ Antonio and Tuffley, "The Gender Digital Divide In Developing Countries."

¹⁶² Interviewee 16

¹⁶³ Interviewee 32

¹⁶⁴ Interviewee 18



VII. CONCLUSION



VII. Conclusion

Globally, there is a persistent gap in MHPSS provision that has become shockingly apparent in the COVID-19 pandemic. Digital technologies have emerged in recent years as an innovative solution to this provision gap, offering solutions to many of the specific needs faced by the Movement in delivering care to populations in emergencies and beyond, such as resource gaps and lack of access.

In light of this potential, the Movement has committed to increasing digital MHPSS in its MHPSS Roadmap 2020-2023 and has started digitalising some MHPSS services. To contribute to these efforts, this report reviewed 37 digital MHPSS interventions (Annex 2) inside and outside the Movement to identify common challenges and practical solutions (Table 2) which were synthesised into a process chart (Figure 10). Our analysis has highlighted that digital technologies are not only useful for the delivery of services to end-users but can be employed throughout the process from engaging with focus groups, supervision, and training, to planning, monitoring, and evaluation.

Based on the above, this report has formulated six overarching recommendations to effectively leverage digital MHPSS tools:

- 1. Collect and leverage data for monitoring, evaluation, and planning
- 2. Favour hybrid models
- 3. Adapt existing evidence-based interventions
- 4. Engage in principled partnerships
- 5. Build synergies with local structures
- 6. Increase offers in training, SOPs, and supervision

Their adoption could mean a significant step forward in providing effective and efficient care. Not least, we hope that the SESSP Analytical Framework outlined in the report will help guide future development and evaluation.

In acknowledgement of the many limitations of this report, we hope that our contribution is the start of a growing dialogue. Further research is needed to consolidate our knowledge of the field. This includes in-depth analyses of technical solutions, reviews of private sector interventions, and precise guidelines on how to structure business-humanitarian partnerships without compromising humanitarian principles, among others. In particular, the Movement could make a highly valuable contribution in growing the evidence base on basic psychosocial interventions in which it is strongly engaged.

Finally, to generate and share further insights on how to improve MHPSS provision, we would encourage the Movement to create a durable platform for knowledge transfer and dialogue within the MHPSS community. We are confident that digital tools will reliably facilitate these exchanges.



BIBLIOGRAPHY



Bibliography

- Aebersold, Monia. "E-Mental-Health für traumatisierte Geflüchtete. Bericht zum Schritt I: Literatur- und Marktanalyse von erprobten und bewährten Materialien und Ansätzen". 2019. Swiss Red Cross.
- American Psychological Association "Understanding Psychological First Aid". 2019. Available at: <u>https://www.apa.org/practice/programs/dmhi/psychological-first-aid</u>.
- Antonio, Amy, and David Tuffley. "The Gender Digital Divide In Developing Countries". *Future Internet* 6, no. 4 (2014): 673-687. doi:10.3390/fi6040673.
- Attridge, Mark. "Internet-Based Cognitive-Behavioral Therapy For Employees With Anxiety, Depression, Social Phobia, Or Insomnia: Clinical And Work Outcomes". *SAGE Open* 10, no. 1 (2020): 215824402091439. doi:10.1177/2158244020914398.
- Bangpan, Mukdarut, Lambert Felix, and Kelly Dickson. "Mental Health And Psychosocial Support Programmes For Adults In Humanitarian Emergencies: A Systematic Review And Meta-Analysis In Low And Middle-Income Countries". *BMJ Global Health* 4, no. 5 (2019): e001484. doi:10.1136/bmjgh-2019-001484.
- Berryhill, M Blake, Anne Halli-Tierney, Nathan Culmer, Nelle Williams, Alex Betancourt, Michael King, and Hannah Ruggles. "Videoconferencing Psychological Therapy And Anxiety: A Systematic Review". *Family Practice* 36, no. 1 (2018): 53-63. doi:10.1093/fampra/cmy072.
- Borghouts, Judith, Elizabeth Eikey, Gloria Mark, Cinthia De Leon, Stephen M Schueller, Margaret Schneider, Nicole Stadnick, Kai Zheng, Dana Mukamel, and Dara H Sorkin. "Barriers To And Facilitators Of User Engagement With Digital Mental Health Interventions: Systematic Review". Journal Of Medical Internet Research 23, no. 3 (2021): e24387. doi:10.2196/24387.
- Bronfenbrenner, Urie. *The ecology of human development: Experiments by nature and design*. Harvard university press, 1979.
- Brown, F. L., K. Carswell, J. Augustinavicius, A. Adaku, M. R. Leku, R. G. White, and P. Ventevogel et al. "Self Help Plus: Study Protocol For A Cluster-Randomised Controlled Trial Of Guided Self-Help With South Sudanese Refugee Women In Uganda". *Global Mental Health* 5 (2018). doi:10.1017/gmh.2018.17.
- Budd, Jobie, Benjamin S. Miller, Erin M. Manning, Vasileios Lampos, Mengdie Zhuang, Michael Edelstein, and Geraint Rees et al. "Digital Technologies In The Public-Health Response To COVID-19". Nature Medicine 26, no. 8 (2020): 1183-1192. doi:10.1038/s41591-020-1011-4.
- Burns, Ryan. "New Frontiers Of Philanthro-Capitalism: Digital Technologies And Humanitarianism". *Antipode* 51, no. 4 (2019): 1101-1122. doi:10.1111/anti.12534.
- Carbonnier, Gilles, and Piedra Lightfoot. "Business In Humanitarian Crises For Better Or For Worse?". In *The New Humanitarians In International Practice: Emerging Actors And Contested Principles*, 1st ed. London: Routledge, 2015. https://doi.org/10.4324/9781315737621.
- Carter, Helena, Ricardo Araya, Kavya Anjur, Davy Deng, and John A. Naslund. "The Emergence Of Digital Mental Health In Low-Income And Middle-Income Countries: A Review Of Recent Advances And Implications For The Treatment And Prevention Of Mental Disorders". *Journal Of Psychiatric Research* 133 (2021): 223-246. doi:10.1016/j.jpsychires.2020.12.016.
- Charlson, Fiona, Mark van Ommeren, Abraham Flaxman, Joseph Cornett, Harvey Whiteford, and Shekhar Saxena. "New WHO Prevalence Estimates Of Mental Disorders In Conflict Settings: A Systematic Review And Meta-Analysis". *The Lancet* 394, no. 10194 (2019): 240-248. doi:10.1016/s0140-6736(19)30934-1.

- Christensen, Helen, Kathleen Griffiths, and Kimberley Evans. *E-Mental Health In Australia: Implications Of The Internet And Related Technologies For Policy*. Canberra: Commonwealth Dept. of Health and Ageing, 2002.
- Council of Delegates of the International Red Cross and Red Crescent Movement, *International Review of the Red Cross* 87, no. 860. Available at: <u>https://international-</u> <u>review.icrc.org/sites/default/files/irrc</u> 860 11.pdf
- Ellis, Louise A., Hanna Augustsson, Anne I. Grødahl, Chiara Pomare, Kate Churruca, Janet C. Long, Kristiana Ludlow, Yvonne A. Zurynski, and Jeffrey Braithwaite. "Implementation Of E-Mental Health For Depression And Anxiety: A Critical Scoping Review". *Journal Of Community Psychology* 48, no. 3 (2020): 904-920. doi:10.1002/jcop.22309.
- Gary, Faye A. "STIGMA: BARRIER TO MENTAL HEALTH CARE AMONG ETHNIC MINORITIES". *Issues In Mental Health Nursing* 26, no. 10 (2005): 979-999. doi:10.1080/01612840500280638.
- GBD 2019 Mental Disorders Collaborators. "Global, Regional, And National Burden Of 12 Mental Disorders In 204 Countries And Territories, 1990–2019: A Systematic Analysis For The Global Burden Of Disease Study 2019". *The Lancet Psychiatry* 9, no. 2 (2022): 137-150. doi:10.1016/s2215-0366(21)00395-3.

Hamilton, Lorna, and Connie Corbett-Whittier. Using Case Study In Education Research. London: Sage, 2012.

- Harper Shehadeh, Melissa Joanne, Jinane Abi Ramia, Pim Cuijpers, Rabih El Chammay, Eva Heim, Wissam Kheir, and Khalid Saeed et al. "Step-By-Step, An E-Mental Health Intervention For Depression: A Mixed Methods Pilot Study From Lebanon". *Frontiers In Psychiatry* 10 (2020). doi:10.3389/fpsyt.2019.00986.
- Henrich, Joseph, Steven J. Heine, and Ara Norenzayan. "The Weirdest People In The World?". *Behavioral And Brain Sciences* 33, no. 2-3 (2010): 61-83. doi:10.1017/s0140525x0999152x.
- Hubley, Sam, Sarah B Lynch, Christopher Schneck, Marshall Thomas, and Jay Shore. "Review Of Key Telepsychiatry Outcomes". *World Journal Of Psychiatry* 6, no. 2 (2016): 269. doi:10.5498/wjp.v6.i2.269.
- IASC Reference Group on Mental Health and Psychosocial Support in Emergency Settings. MENTAL HEALTH AND

 PSYCHOSOCIAL SUPPORT IN EMERGENCY SETTINGS Monitoring And Evaluation With Means Of

 Verification: Version 2.0, 2021. Available at:

 https://interagencystandingcommittee.org/system/files/2021-

 09/%20IASC%20Common%20Monitoring%20and%20Evaluation%20Framework%20for%20Mental%

 20Health%20and%20Psychosocial%20Support%20in%20Emergency%20Settings-

 %20With%20means%20of%20verification%20%28Version%202.0%29.pdf.
- Ideo.Com "Design Kit: The Human-Centered Design Toolkit". 2022. Available at: https://www.ideo.com/post/design-kit.
- Inter-Agency Standing Committee. *Iasc Guidelines On Mental Health And Psychosocial Support In Emergency Settings*. Geneva, 2007. Available at: <u>https://www.who.int/mental health/emergencies/guidelines iasc mental health psychosocial jun e 2007.pdf</u>.
- ICRC *Guidelines On Mental Health And Psychosocial Support*, 2020. Available at: <u>https://www.icrc.org/en/publication/4311-guidelines-mental-health-and-psychosocial-support</u>.
- IFRC "Mental Health Matters: Progress Report on Mental Health and Psychosocial Support Activities".. Available at: <u>https://www.ifrc.org/document/mental-health-matters-progress-report-2021</u>
- IRCRCM, International Red Cross And Red Crescent Movement Commitments On Addressing Mental Health And Psychosocial Needs: A Roadmap For Implementation 2020 – 2023. Geneva, 2019. Available at: https://pscentre.org/wp-content/uploads/2020/06/MHPSS-roadmap-2020-2023.pdf.

- IRCRM, The Working Group of the International Red Cross and Red Crescent Movement Project on Addressing Mental Health and Psychosocial Consequences of Armed Conflicts, Natural Disasters and Other Emergencies. International Red Cross And Red Crescent Movement Policy On Addressing Mental Health And Psychosocial Needs. Geneva, 2019. Available at: <u>https://pscentre.org/wpcontent/uploads/2020/10/CD19-R5-Adopted-MHPSS-need-policy-draft-resolution-FINAL-EN clean.pdf</u>.
- Lal, Shalini, and Carol E. Adair. "E-Mental Health: A Rapid Review Of The Literature". *Psychiatric Services* 65, no. 1 (2014): 24-32. doi:10.1176/appi.ps.201300009.
- Lal, Shalini. "E-Mental Health: Promising Advancements In Policy, Research, And Practice". *Healthcare Management Forum* 32, no. 2 (2019): 56-62. doi:10.1177/0840470418818583.
- Lee, C., A. J. Nguyen, E. Haroz, W. Tol, Y. Aules, and P. Bolton. "Identifying Research Priorities For Psychosocial Support Programs In Humanitarian Settings". *Global Mental Health* 6 (2019). doi:10.1017/gmh.2019.19
- Lin, Lewei (Allison), Danielle Casteel, Erin Shigekawa, Meghan Soulsby Weyrich, Dylan H. Roby, and Sara B. McMenamin. "Telemedicine-Delivered Treatment Interventions For Substance Use Disorders: A Systematic Review". *Journal Of Substance Abuse Treatment* 101 (2019): 38-49. doi:10.1016/j.jsat.2019.03.007.
- Lungu, Anita, Janie Jihee Jun, Okhtay Azarmanesh, Yan Leykin, and Connie E-Jean Chen. "Blended Care-Cognitive Behavioral Therapy For Depression And Anxiety In Real-World Settings: Pragmatic Retrospective Study". *Journal Of Medical Internet Research* 22, no. 7 (2020): e18723. doi:10.2196/18723.
- Madianou, Mirca. "Technocolonialism: Digital Innovation And Data Practices In The Humanitarian Response To Refugee Crises". *Social Media + Society* 5, no. 3 (2019): 205630511986314. doi:10.1177/2056305119863146.
- Margheri, Andrea, Massimiliano Masi, Abdallah Miladi, Vladimiro Sassone, and Jason Rosenzweig. "Decentralised Provenance For Healthcare Data". *International Journal Of Medical Informatics* 141 (2020): 104197. doi:10.1016/j.ijmedinf.2020.104197.
- Meier, Patrick. Digital Humanitarians. Boca Raton, FL: CRC Press, Taylor & Francis Group, 2015.
- Naslund, John A, Kelly A Aschbrenner, Ricardo Araya, Lisa A Marsch, Jürgen Unützer, Vikram Patel, and Stephen J Bartels. "Digital Technology For Treating And Preventing Mental Disorders In Low-Income And Middle-Income Countries: A Narrative Review Of The Literature". *The Lancet Psychiatry* 4, no. 6 (2017): 486-500. doi:10.1016/s2215-0366(17)30096-2.
- O'Brien, Aileen, Rana Fahmy, and Swaran P. Singh. "Disengagement From Mental Health Services". *Social Psychiatry And Psychiatric Epidemiology* 44, no. 7 (2008): 558-568. doi:10.1007/s00127-008-0476-0.
- Patel, Vikram, and Shekhar Saxena. "Transforming Lives, Enhancing Communities Innovations In Global Mental Health". *New England Journal Of Medicine* 370, no. 6 (2014): 498-501. doi:10.1056/nejmp1315214.
- Patel, Vikram, Shekhar Saxena, Crick Lund, Graham Thornicroft, Florence Baingana, Paul Bolton, and Dan Chisholm et al. "The Lancet Commission On Global Mental Health And Sustainable Development". *The Lancet* 392, no. 10157 (2018): 1553-1598. doi:10.1016/s0140-6736(18)31612-x.
- Paul, Alison, Alison Lynch Richard, Sylvain Roy, and Diana Velikonja. "Increasing Psychological Service Access For COVID-19-Impacted Frontline Workers: Canadian Red Cross & Ontario Psychological Association Partnership". Presentation, Canadian Conference on Global Health 2021, 2021.

- Psychosocial Centre International Federation of Red Cross and Red Crescent Societies. *IFRC Monitoring And Evaluation Framework For Psychosocial Support Interventions*. Copenhagen, 2022. Available at: https://pscentre.org/wp-content/uploads/2018/02/Guidance-Note_ME-framework_FINAL.pdf.
- Psychosocial Centre International Federation of Red Cross and Red Crescent Societies. *The MHPSS Framework*, 2022. Available at: <u>https://pscentre.org/what-we-do/the-mhpss-framework/</u>.
- Purgato, Marianna, Alden L Gross, Theresa Betancourt, Paul Bolton, Chiara Bonetto, Chiara Gastaldon, and James Gordon et al. "Focused Psychosocial Interventions For Children In Low-Resource Humanitarian Settings: A Systematic Review And Individual Participant Data Meta-Analysis". *The Lancet Global Health* 6, no. 4 (2018): e390-e400. doi:10.1016/s2214-109x(18)30046-9.
- Ramachandran, Manoharan, Niaz Chowdhury, Allan Third, John Domingue, Kevin Quick, and Michelle Bachler. "Towards Complete Decentralised Verification Of Data With Confidentiality: Different Ways To Connect Solid Pods And Blockchain". *Companion Proceedings Of The Web Conference 2020*, 2020. doi:10.1145/3366424.3385759.
- Riper, Heleen, Gerhard Andersson, Helen Christensen, Pim Cuijpers, Alfred Lange, and Gunther Eysenbach. "Theme Issue On E-Mental Health: A Growing Field In Internet Research". *Journal Of Medical Internet Research* 12, no. 5 (2010): e74. doi:10.2196/jmir.1713.
- Santomauro, Damian F, Ana M Mantilla Herrera, Jamileh Shadid, Peng Zheng, Charlie Ashbaugh, David M Pigott, and Cristiana Abbafati et al. "Global Prevalence And Burden Of Depressive And Anxiety Disorders In 204 Countries And Territories In 2020 Due To The COVID-19 Pandemic". *The Lancet* 398, no. 10312 (2021): 1700-1712. doi:10.1016/s0140-6736(21)02143-7.
- Schneider, Justine, Pooria Sarrami Foroushani, Paul Grime, and Graham Thornicroft. "Acceptability Of Online Self-Help To People With Depression: Users' Views Of Moodgym Versus Informational Websites". *Journal Of Medical Internet Research* 16, no. 3 (2014): e90. doi:10.2196/jmir.2871.
- Shah, Siddharth Ashvin. "Ethical Standards For Transnational Mental Health And Psychosocial Support (MHPSS): Do No Harm, Preventing Cross-Cultural Errors And Inviting Pushback". *Clinical Social Work Journal* 40, no. 4 (2011): 438-449. doi:10.1007/s10615-011-0348-z.
- Shultz, James M, and David Forbes. "Psychological First Aid". *Disaster Health* 2, no. 1 (2013): 3-12. doi:10.4161/dish.26006.
- Snider, Leslie, Mark Van Ommeren, and Alison Schafer. *Psychological First Aid*. Geneva, Switzerland: World Health Organization, (2011).
- Snider, Leslie. Psychological First Aid: Five Year Retrospective (2011-2016). Church of Sweden, 2018. Available at: https://pscentre.org/wp-content/uploads/2021/07/pfa-5-year-retrospective-report-final.pdf
- Steel, Zachary, Claire Marnane, Changiz Iranpour, Tien Chey, John W Jackson, Vikram Patel, and Derrick Silove.
 "The Global Prevalence Of Common Mental Disorders: A Systematic Review And Meta-Analysis 1980–2013". International Journal Of Epidemiology 43, no. 2 (2014): 476-493. doi:10.1093/ije/dyu038.
- The Lancet Global Health. "Mental Health Matters". *The Lancet Global Health* 8, no. 11 (2020): e1352. doi:10.1016/s2214-109x(20)30432-0.
- Tongco, Ma. Dolores C. "Purposive Sampling As A Tool For Informant Selection". *Ethnobotany Research And Applications* 5 (2007): 147. doi:10.17348/era.5.0.147-158.
- Ugochukwu, Obianuju, Nkechi Mbaezue, Saheed Akinmayowa Lawal, Chris Azubogu, Taiwo Lateef Sheikh, and Frédérique Vallières. "The Time Is Now: Reforming Nigeria's Outdated Mental Health Laws". *The Lancet Global Health* 8, no. 8 (2020): e989-e990. doi:10.1016/s2214-109x(20)30302-8.

- UNHCR. Effective inclusion of refugees Participatory approaches for practitioners at the local level. N.d. Available at: <u>https://www.unhcr.org/uk/publications/brochures/61f7be2c4/effective-inclusion-refugees-</u> participatory-approaches-practitioners-local.html?query=participatory
- UNICEF. *Mental Health And Psychosocial Technical Note*. New York, 2019. Available at: https://www.unicef.org/media/73726/file/UNICEF-MH-and-PS-Technical-Note-2019.pdf.pdf.
- van 't Hof, Edith, Eva Heim, Jinane Abi Ramia, Sebastian Burchert, Ilja Cornelisz, Pim Cuijpers, and Rabih El Chammay et al. "Evaluating The Effectiveness Of An E-Mental Health Intervention For People Living In Lebanon: Protocol For Two Randomized Controlled Trials". JMIR Research Protocols 10, no. 1 (2021): e21585. doi:10.2196/21585.
- Venkatesan, Aarathi, Lily Rahimi, Manpreet Kaur, and Christopher Mosunic. "Digital Cognitive Behavior Therapy Intervention For Depression And Anxiety: Retrospective Study". *JMIR Mental Health* 7, no. 8 (2020): e21304. doi:10.2196/21304.
- WHO. "Suicide", 2022. Available at: https://www.who.int/news-room/fact-sheets/detail/suicide.
- WHO. "Mental Health: Strengthening Our Response". 2018. Available at: <u>https://www.who.int/news-room/fact-sheets/detail/mental-health-strengthening-our-response</u>.
- WHO. "The Impact Of COVID-19 On Mental, Neurological And Substance Use Services". 2020. Available at: https://www.who.int/publications/i/item/978924012455.
- WHO. MentalHealthAtlas2020.Geneva,2021.Availableat:https://www.who.int/publications/i/item/9789240036703.



ANNEXES



<u>Annexes</u>

Annex 1: Interviewee List

#	Region	Organisation	Date
1	Europe	WHO	27/01/2022
2	Europe	WHO	28/01/2022
3	Europe	University of Bern	14/02/2022
4	Middle East	UNDP Syria	02/02/2022
5	Europe	Bipolar Buddy	02/02/2022
6	Africa	Psychiatrist	02/02/2022
7	Europe	Swiss Red Cross	26/01/2022
8	Europe	Danish Red Cross	09/02/2022
9	Asia-Pacific	Australian Red Cross	20/01/2022
10	North America	American Red Cross	15/01/2022
11	North America	Global Disaster Preparedness Center (GDPC)	02/02/2022
12	Latin America	IFRC Panama Regional Office	02/03/2022
13	Asia-Pacific	ICRC	13/01/2022
14	Africa	ICRC	19/01/2022
15	Europe	Netherlands Red Cross	22/02/2022
16	Africa	Kenyan Red Cross	23/02/2022
17	Asia -Pacific	New Zealand Red Cross	02/02/2002
18	North America	Canadian Red Cross	16/02/2022
19	Europe	Linköping University	07/02/2022
20	Europe	University of Amsterdam	11/02/2022
21	Europe	IFRC	16/02/2022
22	Latin America	Mexican Red Cross	22/02/2022
23	Latin America	Brazilian Red Cross	15/02/2022
24	Africa	Ethiopian Red Cross Society	29/02/2022
25	Africa	Zambian Red Cross Society	21/02/2022
26	Asia -Pacific	IFRC	15/02/2022
27	Middle East	Medical Corps	15/02/2022
28	Middle East	Medical Corps	22/02/2022
29	Europe	University of Basel	16/02/2022
30	Middle East	Medical Corps	21/02/2022
31	Latin America	Argentinian Red Cross	02/03/2022
32	Africa	Zambian Red Cross Society	22/02/2022

Annex 2: List of Interventions

Name	Technology	Pyrami d Level	Туре	Guided	Target Group	Region	Organisatio n	Partnership s	Description/Object ive	Status	Evaluation
Get Prepared	Mobile App	1	Restoring family links, emergency preparedness	no	General	Australia	Australian Red Cross	/	Reconnect key contacts after an emergency, organisationally and mentally prepare for emergencies	available	/
<u>MyTeam</u>	Mobile App	1	Self-care awareness, peer support	no	General	Australia	Australian Red Cross	Beyond Blue	Organise peer support to achieve set mental health goals	available	/
After the Emergency	Podcast	1	Psychoeducati on / Stress Management	no	General	Australia	Australian Red Cross	Australian Broadcast Company	Podcast with psychology experts to help people recover from traumatic experiences	available	/
After the Emergency	MP3/Websi te	1	Psychoeducati on / Stress Management / Self-care awareness	no	Young people	Australia	Australian Red Cross		Website from which MP3 files containing interviews with material on psychoeducation, stress management and self-awarenesscan be sourced. Distributed on MP3-players in the 2009 bush fires.	suspended	/

Sui	Mobile App / Web App	2/3	CBT, Self-care awareness, stress management, structural help	both	Adult, literate Arabic- speaking refugees	Switzerlan d	Swiss Red Cross	University of Bern; Freie Universität Berlin	Mobile Application that offers practical information on housing, health services, etc. to attract users and then engages them in autonomous or peer guided mental health exercises.	upcoming	2 RCTs
SAHA	Mobile App	3	Cognitive Behavioural Therapy	yes	Arabic refugees suffering depression	Sweden	University of Linköping	/	9 training modules on different mental health topics on which homework is done that is reviewed by practitioners	upcoming	RCT showing positive effect
<u>Self Help</u> <u>Plus (SH+)</u>	Online Training	2	Stress Management, Psychoeducati on	yes	Adults	Global	WHO	/	Guided Workshops of approx. 20 participants	available	3 RCTs, for offline use
Problem Managemen t Plus (PM+)	Online Training	2	Stress Management, Psychoeducati on	yes	Adults	Global	IFRC PS Centre	WHO	Guided Workshops of approx. 20 participants	available	RCTs for offline use
<u>Step-by-</u> <u>Step</u>	Mobile App / Website	2/3	Psychological Intervention for depression	both	People affected by adversities in any contexts with access to the internet	Global	WHO	Egyptian Caritas / National MH Programm e Lebanon, Free University Berlin, University of Zurich, VU	Psychoeducation and training through an illustrated narrative with additional therapeutic techniques such as stress management, identifying strengths, positive	Available (publication expected in 2022) contact psych_interventions@wh o.int	2 fully powered RCTs in Lebanon (papers in preparatio n). 3 RCTs under the STRENGTH S project in Germany, Sweden,
								University Amsterda m, Fondation d'Harcourt, elhra / R2HC	self-talk, increasing social support and relapse prevention		Egypt. Results due in 2022
--------------	--	-----	--	------------------------------------	------------------------------	-----------------	------------------------	---	--	-----------	----------------------------------
STARS	Chatbot	1/2	Psychoeducati on / Stress Management / Self-care awareness	no	Adolescents 15-18	Global	WHO	Fondation Botnar	Chatbot which imitates human interaction to provide adolescent with adequate treatments / exercises according to their reported symptoms	upcoming	/
<u>mhGAP</u>	Mobile App	3	Clinical Intervention Guide	no	Practitioners	Global	WHO	/	Intervention Guide for mental, neurological and substance use (MNS) disorders in non-specialist health settings for integrated management of priority MNS conditions using algorithms for decision making.	available	/
START Now	Mobile App / Website (still piloting)	1	Psychoeducati on / Emotional regulation training	piloting with and without	young people (aged 14-24)	Switzerlan d	University of Basel	Private company, Philanthro pic sponsor	START Now is a skills training to promote resilience and improve stress and emotion regulation. Piloting to turn a paper- and-pencil into web-based training.	upcoming	RCT

Stay fine App	Mobile App	3	Relapse prevention	peer guidanc e	young people (13-21) recovered from anxiety and/or depression	Netherlan ds	University of Amsterdam	Private company	A personalized monitoring and intervention app to prevent relapse of anxiety and depressive disorders with 8 tailored modules	upcoming	/
<u>Universal</u> <u>App</u> <u>Program</u>	Mobile App	/	/	/	General	Global	Global Disaster Preparedne ss Center	/	Efficient and cost- effective access to an easily- adaptable mobile app platform. The GDPC works with NS to facilitate the localization and customization of app content.	available	/
First Aid App	Mobile App	1	Psychological First Aid	No	General	Global	Global Disaster Preparedne ss Center	IFRC PS Centre	Based on the existing First Aid App launched under the Universal App Program, the GDPC is planning to launch an update which includes a section on Psychological First Aid.	upcoming	/
<u>Bipolar</u> <u>Buddy</u>	Mobile App	4	Mood tracking	no	People with bipolar disorder	United Kingdom	Bipolar Buddy	Microsoft Health	Digital guide to track moods, recognise triggers, and identify patterns to gain more control. The intervention complements psychiatric treatments by making the data	upcoming	1

									accessible to clinicians for enhanced diagnosis and treatment		
Boblberg	Mobile App / Website	1	Peer Support / Social Media	no	General	Denmark	Danish Red Cross	Boblberg	Boblberg is simply a digital message board where users can create a "bubble" to seek new friendships, pursue common hobbies, etc., bringing people together in communities and fighting loneliness	available	400,000 users, 90,000 monthly users, Positive surveys
SnakSamme n	(Video)Pho ne Hotline	2	Peer Support	yes	People suffering loneliness	Denmark	Danish Red Cross	/	Lonely people can book video calls with Red Cross volunteers	available	/
Julevenner	Phone hotline	2	Peer Support	yes	People suffering loneliness	Denmark	Danish Red Cross	/	People who are alone at Christmas are matched with people who have an extra seat at the table on Christmas Eve, supported by a call centre with Red Cross volunteers.	available	/
Remote psychiatric therapy	Videocalls	4	Psychiatric treatment	yes	General	General	General	/	Using video call technology to conduct psychiatric treatments, mostly during COVID and as blended	available	/

									therapy (online + offline)		
Tele- counselling	Phone hotline	3	Psychological Counselling	yes	General	Kenya	Kenyan Red Cross	IFRC, Mastercard , Global Fund	Channel for people to seek help/ have an ear	available	/
Mental health chat	Messenger hotline	2	Psychological Counselling	yes	General	Kenya	Kenyan Red Cross	IFRC, Mastercard , Global Fund	Another channel for people to reach out	available	/
Remote supervision	Videocalls	2	Supervision	yes	Practitioners	General	General	/	Using WhatsApp to facilitate peer support and supervision among mental health care workers in Covid- 19 isolation facilities	available	/
Remote supervision	Text Messenger	2	Supervision	yes	Volunteers	Zambia	Zambian Red Cross	Zambian Ministry of Health	Using video call technologies for supervision of mental health care professionals	available	/
Remote group therapy	Videocalls	2/3	Group counselling	yes	Victims of domestic violence	Brazil	Brazilian Red Cross	/	Using video call technology to conduct group therapy. Recreation of in person therapy	available	(Symptoms evolution)
Remote group therapy	Videocalls	2	Group counselling	yes	General	Lebanon	Internation al Medical Corps	/	Using video call to conduct group therapy: awareness session (depression, anxiety)	available	(Client feedback)

Remote psychologica l therapy	Videocalls	3	Counselling	yes	General	Lebanon	Internation al Medical Corps	/	Using video call technologies to conduct psychological therapy	available	/
Remote assitance service (TAD T Telea(sisten cia Domiciliaria)	Mobile App + Phone hotline	2/3	Support network	yes	Elderly (adapted to other groups during COVID-19)	Argentina	Argentinian Red Cross	Telemed care (private company)	Bracelet, or mobile phone application) to put elderly in contact with a Telecare Centre of the Argentine Red Cross with the aim to (a) help the elderly with emergency management, (b) accompany them when they need to talk/feel lonely and (c) help them in their daily activities	available	(12 years of experi- ence)
Peer- support	Mobile phone	2	Peer Support	yes	Volunteers and Staff	Netherlan ds	Netherland s Red Cross	/	Peer support conducted through text messages and phone calls	available	still in pilot stage
WhatsApp helpline	Mobile phone	1	Peer Support	yes	Undocument ed Migrants / General	Netherlan ds	Netherland s Red Cross		Using WhatsApp to facilitate peer support to those in the Netherlands that do not speak Dutch. The service is provided in different languages and ensures confidentiality and anonymity. The main goal is to provide	available	survey

									information about practical questions such as where to get a vaccine		
E-referral system	Database	4	Needs matching		People in hotel quarantine	Canada	Canada Red Cross	Canadian Public health agency / Ontario Psychologis ts Association	The Public Health Agency in Canada and Ontario Psyc Association have digital system of people who need top tier specialized mental health support. The Canadian RC is leveraging those systems, doing matching those who are quarantining with psychologists that suit them	available	conducted by the Ontario Psychologis t Association
PFA remote training	Videocalls	1	PFA Training	yes	Staff and Volunteers in the Pacific Region	Pacific Islands	IFRC (Pacific)	/	Training through video calls on PFA to staff and volunteers of the NS of the Pacific Region	available	/
<u>Fadfada</u>	Website Platform	2/3	Varying from family counseling to psychiatric therapy	yes	Everyone can access it, but focus groups affected by 'double stigma'	Syria	UNDP Syria	WHO	Online Psychosocial support platform providing confidential support through virtual media. Service users	available	client feedback

									register in the platform and after having assessed the risk and the type of needs, UNDP calls back and conducts around 6 sessions through the phone (depending on needs)		
PEARL Database	Database	/	Supervisory tool + needs matching		Clinicians and practitioners	Africa	ICRC	/	Digitalisation and recompilation of medical files in a centralised manner with the goal of improving psychological interventions and supervision, increasing ability to make predictions about needs, and improving accountability towards donors	not available due to technical problems	/
PFA remote training	Videocalls (online webinars) + learning platform online	1	PFA Training and disaster preparedness	yes	NZRC staff and volunteers + General public	New Zealand	New Zealand Red Cross	Spire - online learning company	Using videocalls to organise webinars and train on PFA to volunteers and staff as well as the public. For internal training, these webinars are complemented by materials uploaded to the Spire online- learning platform	available	/

Necesito Un JuveAmigo (#NUJA)	Social Media	1/2	Peer support and counselling	yes	Young people	Mexico	Mexican Red Cross	OPS (WHO), Contacto Joven	Through the hashtag #NUJA, it is possible to identify those in need of MHPSS. Once they have been identified the Mexican Red Cross calls them, assess their needs and supports and counsels them accordingly. In addition, the intervention has created a network of peer support among young	available	
Virtual Reality- based Physical Rehabilitatio n	Virtual Reality Imagery	3	Rehabilitation / Chronic Pain	yes	chronic pain patients	Southeast Asia	IFRC	ICRC	volunteers Virtual reality – guided imagery- based intervention (seeing a waterfall, soothing, voice- over guiding into relaxation), once each day for 10 days. After 10 days users would already associate mental imagery and relaxation with the audio, and can take away audio tape to practise at home	available in the Philippines	anecdotal client feedback

Annex 3: Interview Guide

Best Practices in Digital MHPSS

Interview Guide

Торіс	Interview Q	uestions							
Professional profile	What position do you hold in your organization?								
Project Details	 What dig What was What was target gro Why did y 	ital MHPSS interventions did your organization implement? is the aim of the project? is the context of the intervention (non/emergency setting, localities, pup, size)? you implement this intervention digitally?							
Project Design	 How big is your team? Did you partner with other organizations or outsource tasks? What is the business model / financing scheme of the intervention? 								
Challenges in Implementatio n	 Do you the What chan legal advine What work What work What strain general? 	nink your intervention was successful? If not, why? Illenges did you face during the implementation (IT, data protection, ce, trained staff, availability of participants)? uld you have done differently? uld you have needed to make your interventions even better? uctural needs do you think there are for better digital MHPSS in							
	Effectiveness	 Positive effect on mental health (clinical trials?) Acceptance of intervention by the target population task-transfer 							
Ducient	Scalability	 Accessibility Ease of adaptation Resource-intensiveness 							
Evaluation	Sustainability	Temporal sustainabilityStructural sustainability							
	Safety	 Do no harm Adherence of ethics (data security) Referral system 							
	Participation	InclusivityDiversity							
Wrap-up	 What les Organizat As a memmental he Are you a field we s 	sons learnt do you want to pass on to other National Societies / tions doing digital MHPSS? her of the Movement, would you be interested in joining the digital ealth WK or know more about the pledge? ware of any other intervention of interest/ other stakeholders in the hould talk to? Could you introduce us?							